

GOVERNMENT OF MAHARASHTRA

LAW AND JUDICIARY DEPARTMENT

INDEX TO ACTS AND REGULATIONS IN FORCE IN THE STATE OF MAHARASHTRA

[Corrected upto the 1st June 2019]



PRINTED IN INDIA BY THE GOVERNMENT CENTRAL PRESS, CHARNI ROAD, MUMBAI 400 004 AND PUBLISHED BY THE DIRECTOR, GOVERNMENT PRINTING, STATIONERY AND PUBLICATIONS, MAHARASHTRA STATE, MUMBAI 400 004.

2019

[*Price* : Rs. 104.00]

CONTENTS

Preface		PAGE
Alphabetical List		1
PART I		
State Acts and Regulations in force in the State of Maharashtra		1
PART II		
Local Acts of the Governor-General in Council in force in the State of Maharashtra including (unrepealed) Regulations.	• •	22
PART III		
State Acts amending unrepealed Central Acts in their application to the State of Maharashtra or apply them uniformly to the whole of the State of Maharashtra.	••	24
Subjectwise List		
PART IV		
Subjectwise Index to State Acts and Regulations in force in the State of Maharashtra.	••	30

PREFACE

All Acts and Regulations in force in the State of Maharashtra including the unrepealed Acts and Regulations of the former State of Madhya Pradesh and Hyderabad, which are not unified and are still in force in the Vidarbha and the Hyderabad areas of the State of Maharashtra, are indexed in this publication, both alphabetically and subjectwise. The earlier publication modified upto the 31st March 2015 is now modified upto the 1st June 2019.

2. The Alphabetical list has been divided into three Parts and a separate Part IV has also been made for subjectwise Index, as follows:—

Part I contains a list of Regulations and State Acts, enacted during the period from 1799 to 1st June 2019 and in force in the State of Maharashtra.

Part II contains a list of local Acts of the Governor-General in Council in force in the State of Maharashtra. It also includes unrepealed Regulations.

Part III contains a list of State Acts, which amend unrepealed Central Acts in their application to the State of Maharashtra, or apply them uniformly to the whole of the State of Maharashtra.

Part IV contains Subjectwise Index of State Acts and Regulations in force in the State of Maharashtra.

MANTRALAYA,

Secretary to Government,

Dated: 1st June 2019.

Law and Judiciary Department.

Mumbai.

ALPHABETICAL LIST OF STATE ACTS AND REGULATIONS IN FORCE IN THE STATE OF MAHARASHTRA

PART I

Note. - Abbreviations used in this Index :

H.C. = Hyderabad Code.

M.P.C. = Madhya Pradesh Code.

M.C. = Maharashtra Code (contains Acts from the year 1827 onwards).

U.C. = Unrepealed Central Acts (1950 Ed.).

STATE ACTS AND REGULATIONS

Title of the Act	Year	Number of the Act
A		
Aadhaar (Targeted Delivery of Financial and other Subsidies, Benefits and Services) Act, Maharashtra.	2017	XVIII
Abkari Act, Bombay.	1878	V
Abkari (Amendment) Act, Bombay.	1912	XII
Abolition of Cash Grants (Hyderabad) Act.	1952	XXXIII
Abolition of Inams (Hyderabad) Act.	1955	VIII
Abolition of Inams (Amendment) Act, Hyderabad.	1959	LXIV
Abolition of Jagirs Regulation (Hyderabad).	1358F	LXIX
Abolition of Whipping Act, Bombay.	1957	XXXIX
Abolition of Proprietory Rights (Estates, Mahals, Alienated Lands) Act (M. P.).	1951	I
Abolition of Subsisting Proprietary Rights to Mines and Minerals in certain Lands Act, 1985—Maharashtra.	1985	XVI
Acupuncture System of Therapy Act, Maharashtra.	2017	XXV
Act for Avoiding Wagers (Amendment) Act, Bombay.	1865	III
Advocates Welfare Fund Act, Maharashtra.	1981	LXI
Adjudication Proceedings (Transfer and Continuance) Act, Bombay.	1947	XXXIV
Adjustment and Liquidation of Industrial Workers' Debt Act (C. P. & Berar).	1936	V
Aerial Ropeways Act, Maharashtra.	1956	III
Agricultural Debtors Relief Act, Maharashtra.	1947	XXVIII
Agricultural Debtors Relief (Hyderabad) Act.	1956	XVI
Agricultural Debtors Relief (Mehwassi Estates and Akrani Mahal) (Amendment) Regulation, Bombay.	1952	Bom. Reg. 11
Agricultural Income-tax Act, Maharashtra .	1962	XLI
Agricultural Income-tax (Validity of Notices) Act (Hyderabad).	1954	II
Agricultural Improvement Fund Act (Hyderabad).	1952	XLV

Title of the Act	Year	Number of the Act
Agricultural Lands (Ceiling on Holdings) Act, Maharashtra.	1961	XXVII
Agricultural Pests and Diseases Act, Maharashtra.	1947	XLIII
Agricultural Produce Marketing (Regulation and Development) Act, Maharashtra.	1964	XX
Agricultural Produce Market Act (C. P. & Berar).	1935	XXIX
Agricultural Universities (Krishi Vidyapeeths) Act, Maharashtra.	1983	XLI
Akrani Mahal (Application of Laws) Regulation.	1952	Reg. I
Alluvion and Dilluvion Regulation (Bengal).	1825	Reg. XI
Amity University Act.	2014	XIII
Anatomy Act, Maharashtra.	1949	XI
Ancient Monuments and Archaeological Sites and Remains Act, Maharashtra.	1961	XII
Animal and Fishery Sciences University Act, Maharashtra.	1998	XVII
Animal Preservation Act, Maharashtra	1977	IX
Apartment Ownership Act, Maharashtra.	1971	XV
Application of Central Acts (Hyderabad).	1952	XLVIII
Atiyat Enquiries (Hyderabad) Act.	1952	X
Attached Estates Management Regulations (Bengal).	1827	Reg. V
Authoritative Text Regulation, Hyderabad.	1358F	XVII
В		
Baronetcies (Amendment) Act, Maharashtra.	1974	XXVIII
Bhagdari and Narwadari Tenures Abolition Act, Maharashtra.	1949	XXXII
Bhagelas Contracts Act.	1353F	IX
Bhil Naik Inams Abolition Act, Maharashtra.	1955	XXI
Bhudan Yagna Act (M. P.).	1953	XV
Bandhijama Udhad and Ugadia Tenures Abolition Act, Maharashtra.	1959	XXXV
Bombay (Okhamandal Salami Tenure Abolition) Act.	1954	I
Bombay City (Inami and Special Tenures) Abolition and Maharashtra Land Revenue Code (Amendment) Act.	1969	XLIV
Bombay Municipal (Extension of Limits) Act.	1950	VII
Bombay University (Temporary Postponement of Election of University Representatives) Act.	1989	XL
Borstal Schools Act, Maharashtra.	1929	XVIII
Borstal Schools (Extension and Amendment) Act, Bombay.	1960	XXI
Building (Control on Erection, Re-erection and Conversion) (Repeal) Act, Bombay.	1971	XLI
Building (Control on Erection) Regulation, Bombay.	1950	Bom. Reg. I

Title of the Act	Year	Number of the Act
С		
Casinos (Control and Tax) Act, Maharashtra.	1976	XXXI
Central India Spinning, Weaving and Manufacturing Company Limited, the Empress Mills, Nagpur (Acquisition and Transfer of Undertaking) Act.	1986	XLVI
Central Provinces and Berar Finance (Bombay Repeal) Act.	1958	XLVII
Change of Short Titles of Certain Bombay Acts.	1980	XV
Change of Short Titles of Certain Bombay Acts.	2012	XXIV
Charged Expenditure Act, Maharashtra.	1957	V
Chhatrapati Shivaji Maharaj Vastu Sangrahalaya (Museum) Act.	1909	III
Chatrapati Shivaji Maharaj University, Panvel, Act, Maharashtra.	2018	XXXII
Children Protection Act (Hyderabad).	1343F	IX
Cinemas (Regulation) Act, Maharashtra.	1953	XI
City Civil Court Act, Bombay.	1948	XL
City of Bombay (Building Works Restriction) Act.	1949	XLIV
Civil Jails Act.	1874	II
Collection of Statistics Act (C. P. & Berar).	1939	XXXIV
College Teacher's (Temporary Postponement of Confirmation) Act, Maharashtra.	1973	XXXIII
Commissioners of Divisions Act, Maharashtra.	1958	VIII
Commissioners of Police Act, Maharashtra State.	1959	LVI
Contingency Fund Act, Maharashtra.	1956	XLVI
Construction of References to "District Magistrate" in the Hyderabad Area (Repeal) Act, Maharashtra.	1960	XIV
Control of Organised Crime Act, Maharashtra.	1999	XXX
Co-operative Societies Act, Maharashtra.	1961	XXIV
Co-operative Societies (Postponement of Election due to Emergency) Act, Maharashtra.	1972	VIII
Co-operative Societies (Amendment of Liquidators' Orders Validation) Act, (C. P. & Berar).	1945	X
Co-operative Sugar Factories under erection (Postponement of Elections) Act, Maharashtra.	1974	LIX
Co-operative Sugar Factories (under erection and certain others) (Postponement of Elections) Act, Maharashtra.	1977	XLIX
Co-operative Societies (Reconstitution and Formation) Act, (M.P.).	1956	XVIII

Title of the Act	Year	Number of the Act
Co-operative Land Mortgage Banks Act (C. P. & Berar).	1937	I
Coroners Act, (Maharashtra Repeal).	2000	VIII
Cotton Contracts Act, Bombay.	1932	IV
Cotton Contracts Control (War Provisions) Repeal Act, Bombay.	1922	III
Cotton Control Act, Bombay.	1942	XXX
Cotton Seeds (Regulation of supply, distribution, sale and fixation of sale price) Act, Maharashtra.	2009	XIX
Cotton (Statistics) Act, Bombay.	1946	XXVII
Cotton Ginning and Pressing Factories (Hyderabad Amendment) Act.	1954	XXXIII
Cotton Ginning and Pressing Factories (Maharashtra Provision for Uniform Application and Amendment) Act.	1962	I
Court-fees Act, Maharashtra.	1959	XXXVI
Special Provision for Payment of Court Fees Act, Maharashtra.	1976	XXXII
Court of Wards Act (Hyderabad).	$1350\mathrm{F}$	XII
Court of Wards Act (C. P. & Berar).	1899	XXIV
D		
Debt Conciliation Act (C. P. & Berar).	1933	II
Debt Relief Act, Maharashtra.	1976	III
Deletion of the Term "famine" (from Laws applicable to the State) Act, Maharashtra.	1963	XL
Departmental Inquiries (Enforcement of Attendance of Witnesses and Production of Documents) Act, Maharashtra.	1986	XXIX
Destruction of Useless Records Act (Hyderabad).	1305 F	III
Devadasi System (Abolition)Act, Maharashtra.	2006	XXXIII
Director of Prohibition and Excise (Change of Designation) Act, Maharashtra.	1973	LII
Diseases of Animals Act, Bombay.	1948	LIX
Disposition of Property (Bombay) Validation Act.	1947	LIV
Dissolution of Osmanabad and Parbhani Zilla Parishads and Temporary Postponement of Elections Act.	1980	XIX
District Officers (Change of Designation and Construction of References) Act (Hyderabad),	1950	XXXV
District Planning Committees (Constitution and Functions) Act, Maharashtra.	1998	XXIV
Domestic Workers Welfare Board Act, Maharashtra.	2009	I
Drinking Water Supply Requisition Act, Maharashtra.	1983	XX
Dr. Ajeenkya D. Y. Patil University Act, Maharashtra.	2015	III

Title of the Act	Year	Number of the Act
Dr. Babasaheb Ambedkar Technological University Act.	2014	XXIX
Dr. Vishwanath Karad MIT World Peace University Act, Maharashtra.	2017	XXXV
Drugs (Control) Act, Maharashtra .	1960	XI
Drugs (Control) Act (C. P. & Berar).	1949	XLVII
DSK World University Act, Maharashtra.	2017	XXXIX
D.Y. Patil International University, Pune, Act, Maharashtra.	2017	LXIII
Education and Employment Guarantee (Cess) Act, Maharashtra.	1962	XXVII
Educational Institutions (Management) Act, Maharashtra.	1976	XIII
Educational Institutions (Prohibition of Capitation Fee) Act, Maharashtra.	1988	VI
Educational Institutions (Regulations of Fee) Act, Maharashtra.	2014	VII
Educational Institutions (Transfer of Management) Act, Maharashtra.	1971	XLIX
Edulabad and Warangaon Parganas Laws Act.	1866	XIV
Electricity Duty Act, Maharashtra.	2016	XXVI
Electricity (Special Powers) Act, Maharashtra.	1946	XX
Electricity (Special Powers) Act, (Application to Scheduled Areas) Regulation, Bombay.	1955	Reg. I
Employees of Private Schools (Conditions of Service) Regulation Act, Maharashtra.	1978	III
Employment Guarantee Act, Maharashtra	1978	XX
Entertainments Duty Act, Maharashtra	1923	I
Eradication of Unfair Activities at Pandharpur Temples (for providing better facilities for worship) Act, 1980.	1980	XI
Essential Commodities and Cattle (Control) Act, Bombay.	1958	LXII
Essential Services Maintenance Act, Maharashtra.	2018	XVIII
Evacuee Interest (Separation) Supplementary Act (Hyderabad).	1953	VII
Excise Act (C. P. & Berar).	1915	II
Execution of Decrees (Temporary Postponement) Act, Bombay.	1959	LXX
Exemptions from Land-Revenue (No. 1) Act.	1863	II
Exemptions from Land-Revenue (No. 2) Act.	1863	VII
Extension of Municipal Limits (Validation) Act, Maharashtra. ${f F}$	1967	V
Felling of Trees (Regulation) Act, Maharashtra.	1964	XXXIV
Ferries and Inland Vessels Act, Maharashtra.	1868	II

Title of the Act	Year	Number of the Act
Finance (Bombay Repeal) Act, C. P. and Berar.	1958	XLVII
Finance Commission (Miscellaneous Provisions) Act, Maharashtra.	1994	XXIII
Fire Prevention and Life Safety Measures Act, Maharashtra.	2007	III
Fiscal Responsibility and Budgetary Management Act, Maharashtra.	2005	XVI
Fisheries Act, Maharashtra.	1961	I
Flame University Act, Maharashtra.	2015	II
Fodder and Grain Control Act, Bombay.	1939	XXVI
Foreign Immigrants Regulation (Bengal).	1812	XI
Forest Produce (Regulation of Trade) Act, Maharashtra.	1969	LVII
Forward Contracts Control Act, Maharashtra.	1947	LXIV
Fruit Nurseries (Regulation) Act, Maharashtra.	1969	XLV
${f G}$		
Game Act (C. P. & Berar).	1935	XV
Gandhi National Memorial Fund (Local Authorities Donations) Act.	1953	XVII
Gas Companies Act.	1863	V
Gas Supply Act, Bombay.	1939	IX
General Clauses Act (Hyderabad).	1308 F	III
General Clauses (Amendment and Supplementary) Act, (Hyderabad).	1950	XVIII
General Clauses Act, Maharashtra.	1904	I
Godawari Marathwada Irrigation Development Corporation Act, Maharashtra.	1998	XXIII
Goods and Services Tax Act, Maharashtra.	2017	XLIII
Goods and Services Tax (Compensation to Local Authorities) Act, Maharashtra.	2017	XLI
Government Demands Act (Hyderabad).	1308 F	IV
Government Premises (Eviction) Act, Maharashtra.	1956	II
Government Servants Inquiries (Evidence of Corruption) Act, Maharashtra.	1965	XLIV
Government Servants Regulation of Transfers and Prevention of Delay in Discharge of Official Duties Act, Maharashtra.	2006	XXI
Gramdan Act, Maharashtra.	1965	XXIII
Grants-in-aid to Local Bodies (Repeal) Act, Central Provinces and Berar.	1966	XVI
Greater Bombay Laws and the Bombay High Court (Declaration of Limits) Act.	1945	XVII

Title of the Act	Year	Number of the Act
Greater Bombay Laws and the Bombay High Court (Declaration of Limits)(Amendment) Act.	1950	VIII
G. H. Raisoni University, Amravati, Act, Maharashtra.	2018	XXIX
Ground Water (Development and Management) Act, Maharashtra.	2013	XXVI
Gunthewari Developments (Regularisation, Upgradation and Control) Act, Maharashtra.	2001	XXVII
Н		
Habitual Offenders Act, Bombay.	1959	LXI
Hereditary Offices Act, Maharashtra.	1874	III
Hereditary Offices (Amendment) Act.	1886	V
Highways Act, Maharashtra.	1955	LV
Hindu Heirs Relief Act, Bombay.	1866	VII
Hindu Places of Public Worship (Entry Authorisation) Act, Maharashtra.	1956	XXXI
Hindu Widows' Marriage Act (Hyderabad).	$1347~\mathrm{F}$	I
Hindu Women's Rights to Property (Extension to Agricultural Land) Act, Bombay.	1947	XIX
Highway Act (C. P. & Berar).	1936	XXXIV
High Court (Abolition of Original Jurisdiction) Act (Hyderabad).	1951 F	XII
Home Guards Act, Maharashtra.	1947	III
Homoeopathic Practitioners' Act, Maharashtra.	1960	XII
Horticulture Development Corporation Act, Maharashtra.	1988	XXV
High Court (Hearing of Writ Petitions by Division Bench and Abolition of Letters Patent Appeals) Act, Maharashtra.	1986	XVII
Houses (Rent, Eviction and Lease) Control Act, (Hyderabad).	1954 F	XX
Housing and Area Development Act, Maharashtra.	1977	XXVIII
I		
Improvement Trust Act (Nagpur).	1936	XXXVI
Improvement Trust Tribunal (Validation of Proceedings) Act (Nagpur).	1945	III
Increase of Land Revenue and Special Assessment Act, Maharashtra.	1974	XX
Industrial Development Act, Maharashtra.	1962	III
Industrial Relations Act, 1946, Maharashtra.	1947	XI
Infectious Diseases (Hyderabad) Act.	1950 F	XII
Inferior Village Watans Abolition Act, Maharashtra.	1959	I
Infrastructure Development Enabling Authority Act, Maharashtra.	2018	XXVIII

Title of the Act	Year	Number of the Act
Insolvency Act (Hyderabad).	1351 F	VIII
Invalidation of Hindu Ceremonial Emolument Act.	1926	XI
Irrigation Act, Maharashtra.	1976	XXXVIII
J		
Jagirs (Commutation) Regulation (Hyderabad).	$1359\mathrm{F}$	XXV
Jagirdars Debt Settlement (Hyderabad) Act.	1952	XII
Jeevan Authority Act, Maharashtra.	1976	XLVIII
Judicial Proceedings (Regulation of Reports) Act, Bombay.	1955	XVIII
Juvenile Smoking Act (C. P. & Berar).	1929	VIII
K		
Kauli and Katuban Tenures (Abolition) Act, Maharashtra.	1953	XLIV
Kavi Kuluguru Kalidas Sanskrit Vishavavidyalaya (University) Act.	1997	XXXIII
Keeping and Movement of Cattle in Urban Areas (Control) Act, Maharashtra.	1976	XLV
Khadi and Village Industries Act, Maharashtra.	1960	XIX
Khadi and Village Industries Boards Act (M. P.).	1956	IX
Khar Land Development Act, Maharashtra.	1979	XI
Khoti Abolition Act, Maharashtra.	1950	VI
Konkan Irrigation Development Corporation Act.	1998	III
Krishna Valley Development Corporation Act, Maharashtra.	1996	XV
${f L}$		
Labour Welfare Fund Act, Maharashtra .	1953	XL
Land Alienation Act (C. P. and Berar).	1916	II
Land Acquisition Officers Proceedings Validation Act, Maharashtra.	1949	XXXV
Land Improvement Schemes Act, Maharashtra.	1942	XXVIII
Land Requisition Act, Maharashtra.	1948	XXXIII
Land Revenue (No. 1) Act, Exemptions from.	1863	II
Land Revenue (No. 2) Act, Exemptions from.	1863	VII
Land Revenue Code, Maharashtra.	1966	XLI
Land Revenue Code (Amalgamation of Bombay and Konkan Divisions) Act, Maharashtra.	1983	XXXIX
Land Revenue (Revival of Certain Rules relating to Non-agricultural Assessment) Act, Maharashtra.	1972	XXV
Land (Special Assessment) Act, (Hyderabad).	1952 F	XXXII
Land Survey Act (C. P. and Berar).	1947	XLII

Title of the Act	Year	Number of the Act
Land Tenures Abolition (Recovery of Records) Act, Maharashtra.	1953	L
Land Tenures Abolition (Amendment) Act, Bombay.	1953	XXXVIII
Land Tenures Abolition (Amendment) Act, Bombay.	1956	XL
Land Tenures Abolition Laws (Amendment) Act, Bombay.	1958	LVII
Languages of Judgment and Orders Act (Hyderabad).	1950	XXIX
Laws Act (Central Provisions).	1875	XX
Laws Provincial Act (Berar).	1941	XV
Labour Housing Act (Hyderabad).	1952	XXXVI
Leaders of Opposition in Maharashtra Legislature (Salaries and Allowances) Act.	1978	VIII
Legislative Council (Chairman and Deputy Chairman) and the Bombay Legislative Assembly (Speaker and Deputy Speaker) Salaries and Allowances Act, Maharashtra.	1956	XLVII
Legislature Members (Removal of Dis qualifications) Act, Maharashtra.	1956	LII
Legislature Members' Salaries and Allowances Act, Maharashtra.	1956	XLIX
Legislature Members' Pension Act, Maharashtra.	1977	I
Lifts, Escalators and Moving Walks Act, Maharashtra.	2018	XV
Livestock Improvement Act, Maharashtra.	1933	XXII
Local Authorities (Electoral Offences) Act (M. P.).	1952	XI
Local Authorities Census Expenses Contribution Act, Bombay.	1950	XXIII
Local Authorities (Postponement of Elections for the Duration of the Emergency) (Repeal) Act, Maharashtra.	1963	XXII
Local Authority Members' Disqualification Act, Maharashtra.	1987	XX
Local Bodies Services Commission Act (M.P.).	1950	XXXVII
Local Fund Audit Act, Maharashtra.	1930	XXV
Local Government Service (Declaration as State Civil Service) (Hyderabad) Act.	1956	XX
Lokayukta and Upa-Lokayuktas Act, Maharashtra.	1971	XLVI
Lotteries (Control and Tax) and Prize Competition (Tax) Act, Maharashtra.	1958	LXXXII
Luxury-cum-Entertainment and Amusement Tax on Holders of Television Sets Act, Maharashtra.	1982	XIX
${f M}$		
Marine Fishing Regulation Act, Maharashtra.	1981	LIV
Marine Fishing Services Act, Maharashtra.	2015	XXX
Mahatma Phule Vastu Sangrahalaya, Pune Act.	1968	XVIII

Title of the Act	Year	Number of the Act
Mahul Creek (Extinguishment of Rights) Act.	1922	IV
Mamlatdars' Court Act.	1906	II
Management of Irrigation System by Farmers Act, Maharashtra.	2005	XXIII
Markets and Fairs Act.	1862	IV
Maritime Board Act, Maharashtra.	1997	XV
Marriages Validating Act (C. P. and Berar).	1949	II
Mathadi, Hamal and other Manual Workers (Regulation of Employment and Welfare) Act, Maharashtra.	1969	XXX
Medical and Dental Colleges Admissions(Regulation and Abolition of All India Quota) Act, Maharashtra.	2003	XV
Medical Council Act, Maharashtra.	1965	XLVI
Medical Practitioners Act, Maharashtra.	1961	XXVIII
Medicare Service Persons and Medicare Service Institutions (Prevention of Violence and Damage or Loss to Property) Act, Maharashtra.	2010	XI
Merged Areas, Enclaves and Specified Areas (Amendment of Laws) Act, Bombay.	1950	XXII
Merged States (Laws) Act, Maharashtra.	1950	IV
Merged Territories Revenue Laws (Exercise of Powers) Act, (M.P.).	1951	XX
Merged Territories (Janjira and Bhor) Khot Tenure Abolition Act, Maharashtra.	1953	LXXI
Merged Territories and Areas (Jagirs Abolition) Act, Maharashtra.	1954	XXXIX
Merged Territories Miscellaneous Alienations Abolition Act, Maharashtra.	1955	XXII
Metropolitan Planning Committees (Constitution and Functions) (Continuance of Provisions) Act, Maharashtra.	2000	V
Metropolitan Region Development Region Development Authority Act, Maharashtra.	2017	III
Metropolitan Region Specified Commodities Markets (Regulation of Location) Act, Maharashtra.	1983	XLII
Minister's Salaries and Allowances Act, Maharashtra.	1956	XLVIII
Mineral Development (Creation and Utilisation) Fund Act, Maharashtra.	2001	XIX
Minor Forest Produce (Regulation of Trade) Act, Maharashtra.	1969	LVII
Miscellaneous Alienations (in Hyderabad Enclaves) Abolition Act, Maharashtra.	1965	LVI
MIT Art, Design and Technology University Act, Maharashtra	2015	XXXIX
Molasses (Control) Act, Maharashtra.	1956	XXXVIII

Title of the Act	Year	Number of the Act
Money-Lending (Regulation) Act, Maharashtra.	2014	VIII
Motor Vehicles Tax Act, Maharashtra .	1958	LXV
Maharashtra Requisitioning and Control of Motor Vehicles Act.	1966	II
Motor Vehicles (Taxation of Passengers) Act, Maharashtra.	1958	LXVII
Motor Vehicles (Taxation of Passengers) (Amendment and Validation) Act, 1981 Maharashtra.	1982	IV
Mumbai Metropolitan Region Development Authority Act, Maharashtra.	1974	IV
Municipal Councils, Nagar Panchayats and Industrial Townships Act, Maharashtra.	1965	XL
Municipal Corporation Act, Bombay.	1888	III
Municipal Corporation Act (Hyderabad).	1956 F	II
Municipal Corporations and Municipal Councils (Amendment) Act, Maharashtra.	1993	XV
Municipal Corporations and Municipal Councils (Amendment) Act, Maharashtra.	1994	XLI
Municipal Corporations and Municipal Councils, Nagar Panchayats and Industrial Townships (Amendment) Act, Maharashtra.	1994	XLIV
Municipal Corporation (Bombay) (Revision of Election Rolls and further extension of term) Act.	1978	XVII
Municipal Corporations Temporary Postponement of Elections (of the Mayors and Deputy Meyors due to ensuing general elections to the State Legislative Assembly and certain other elections) Act, Maharashtra.	2004	XVIII
Municipal Councils (Postponement of Elections due to Scarcity Conditions in the State) Act, Maharashtra.	1972	XLVI
Municipal Councils and Municipal Corporations Postponement of Election during the Emergency Act, Maharashtra.	1975	LV
Municipal Councils and Municipal Corporations (Postponement of Elections due to ensuing General Elections to the State Legislative Assembly) Act.	1977	XLVIII
Municipalities (Postponement of Elections due to ensuing General Elections to Municipal Councils) Act, Maharashtra.	1978	XXVI
Municipal Investments Act, City of Bombay.	1898	I
Municipal Property Tax Board Act, Maharashtra.	2011	XIV
Municipal Servants Act, Bombay.	1890	V
N		
National Law University Act, Maharashtra.	2014	VI
Non-Agriculturists' Loans Act, Bombay.	1928	III
Non-biogradable Garbage (Control) Act, Maharashtra.	2006	X

Title of the Act	Year	Number of the Act
Non-trading Corporations Act, Maharashtra.	1959	XXVI
Nurses Act, Maharashtra.	1966	XL
Nursing Home Registration Act, Maharashtra.	1949	XV
O		
Official Languages Act (M.P.).	1950	XXIV
Official Languages Act, Maharashtra.	1965	V
Opium Smoking Act, Maharashtra.	1936	XX
Ownership Flats (Regulation of the promotion of construction, sale, management and transfer) Act, Maharashtra.	1963	XLV
P		
Pandharpur Development Authority Act.	2009	XV
Pandharpur Temples Act.	1974	IX
Pargana and Kulkarni Watans (Abolition) Act, Maharashtra.	1950	LX
Paramedical Council Act, Maharashtra.	2016	VI
Peint Laws Act.	1894	II
Personal Inams Abolition Act, Maharashtra.	1953	XLII
Pleaders Act, Bombay.	1920	XVII
Police Act, Maharashtra.	1951	XXII
Police Officers (Change in Designation) Act, Maharashtra.	1962	XLVI
Poona Mutha River Flood Limits (Prohibition of Building and Provision for Alternative Building Sites) (Repeal) and IndemnityAct.	1966	X
Power Alcohole Act (Hyderabad).	1350 F	XI
Pre-School Centres (Regulation of Admission) (Repeal) Act, Maharashtra.	2001	XXXI
Presidency Small-Causes Courts (Maharashtra Amendment) Act	. 1976	XIX
Prevention and Eradication of Human Sacrifice and other Inhuman, Evil and <i>Aghori</i> Practices and Black Magic Act, Maharashtra.	2013	XXX
Prevention of Begging Act, Maharashtra.	1960	X
Prevention of Adulteration Act, (C. P. and Berar).	1919	II
Prevention of Cruelty to Animals Act, (Hyderabad).	1313 F	I
Prevention of Communal, Anti-social and other Dangerous Activities Act, Maharashtra.	1981	VII
Prevention of Dangerous Activities of Slum-lords, Bootleggers and Drug Offenders Act, Maharashtra.	1981	LV
Prevention of Defacement of Property Act, Maharashtra.	1995	VIII
Prevention of Fragmentation and Consolidation of Holdings Act, Maharashtra.	1947	LXII

Title of the Act	Year	Number of the Act
Prevention of Gambling Act, Maharashtra.	1887	IV
Prevention of Malpractices at University, Board and Other Specified Examinations Act, Maharashtra.	1982	XXXI
Prevention of Thefts from Gold Mines Regulation (Hyderabad).	1359 F	XXIII
Prevention of Adulteration Act (C. P. and Berar).	1919	II
Preventive Detention Act, Maharashtra.	1970	XVIII
Prince of Wales Museum (Amending) Act.	1913	VII
Private Forests Acquisition Act, Maharashtra.	1975	XXIX
Private Professional Educational Institutions (Reservation of Seats for admission for Scheduled Castes, Scheduled Tribes, De-notified Tribes (Vimukta Jatis), Nomadic Tribes and Other Backward Classes) Act, Maharashtra.	2006	XXX
Private Security Guards (Regulation of Employment and Welfare) Act, Maharashtra.	1981	LVIII
Prohibition of Obscene Dance in Hotels, Restaurants and Bar Rooms and Protection of Dignity of Women (Working therein) Act, Maharashtra.	2016	XII
Probation of Offenders Act (C. P. and Berar).	1936	I
Probation of Offenders (Hyderabad) Act.	1953	XII
Prohibition Act, Maharashtra.	1949	XXV
Prohibition of Ragging Act, Maharashtra.	1999	XXXIII
Prohibition of Simultaneous Membership Act, Bombay.	1957	XII
Project Affected Persons Rehabilitation Act, Maharashtra.	2001	XI
Protection of Debtors Act (C. P. and Berar).	1937	IV
Protection of People from Social Boycott (Prevention, Prohibition and Redressal) Act, Maharashtra.	2017	XLIV
Protection of Interest of Depositors (in Financial Establishments) Act, Maharashtra.	2000	XVI
Protection of Pilgrims Act.	1887	II
Protection of Scheduled Tribes (Interest in Trees) Act (M.P.).	1956	XI
Provincial Municipal Corporations Act, Maharashtra.	1949	LIX
Provincial Municipal Corporations and City of Nagpur Corporation (Amendment) Act, Bombay.	1965	XXXIV
Provincial Municipal Corporations and City of Nagpur Corporation (Amendment) Act, Bombay.	1979	XXI
Provincial Municipal Corporations, City of Nagpur Corporation and Maharashtra Municipalities (Amendment) Act, Bombay.	1981	XII
Provincial Municipal Corporations, City of Nagpur Corporation and Maharashtra Municipalities (Second Amendment) Act, Bombay.	1981	LXVIII

Title of the Act	Year	Number of the Act
Provincial Municipal Corporations, City of Nagpur Corporation and Maharashtra Municipalities (Third Amendment) Act.	1981	LXIX
Provincial Municipal Corporations, City of Nagpur Corporation and Maharashtra Municipalities (Amendment) Act, Bombay.	1983	XXVII
Provincial Municipal Corporations and City of Nagpur Corporation (Amendment) Act, Bombay.	1987	XXXVIII
Provisional Collection of Taxes Act, Maharashtra.	1963	V
Provision of Facilities for Agricultural Credit by Banks Act, Maharashtra.	1975	V
Public Authorities Seals Act, Maharashtra.	1883	V
Public Conveyances Act, Maharashtra.	1920	VII
Public Libraries Act, Maharashtra.	1967	XXXIV
Public Libraries (Repeal) Act, Hyderabad.	1961	XL
Public Records Act, Maharashtra.	2006	IV
Public Services (Subordinate) Selection Boards (Repeal) Act, Maharashtra.	1983	XXXIV
Public Trusts Act, Maharashtra.	1950	XXIX
Public Trusts (Unification and Amendment) Act, Bombay.	1960	VI
Public Trusts Act, (M. P.).	1951	XXX
Public Trusts (Validating Provisions and Amendment) Act, Bombay.	1977	XXXV
Pulgaon Cotton Mills Limited (Acquisition of Shares) Act.	1982	XXXIV
${f Q}$		
Queen Victoria Statue Site (and adjoining land tilization for construction of Satellite Telecommunications Exchanges of the Overseas Communications Service) Act, Bombay.	1968	V
${f R}$		
Race-courses Licensing Act, Maharashtra.	1912	III
Railways, Government Irrigation Sources and other Public Buildings Protection Act, (Hyderabad).	1321 F	I
Rajiv Gandhi Science and Technology Commission Act, Maharashtra.	2004	XV
Rationing (Preparatory and Continuance) Measures Act, Bombay.	1947	LVIII
Raw Cotton (Procurement, Processing and Marketing) Act, Maharashtra.	1971	XLVII
Reclamation of Lands (Eradication of Kans) Act, (C. P.).	1948	LXVII
Recognition of Trade Unions and Prevention of Unfair Labour Practices Act, Maharashtra.	1972	I

Title of the Act	Year	Number of the Act
Recovery and Penalty Provisions of Maharashtra Ordinance No. III of 1975 Continuance Act.	1975	XXXIX
Recovery of Arrears of Land Revenue and Rent Act (M. P.).	1953	V
Recognised Examinations Act (C. P. and Berar).	1937	X
Refugees Rehabilitation (Loans) (Maharashtra Repeal) Act (C. P. and Berar).	1963	XI
Refugees Registration and Movement Act (C.P. and Berar).	1947	XXIX
Regional and Town Planning Act, Maharashtra.	1966	XXXVII
Registration of Birth and Deaths Regulations.	1359 F	II
Registrar General of Births, Deaths and Marriage Act (Hyderabad).	1953	VIII
Regulations of Letting of Accomodation Act (C.P and Berar).	1946	XI
Regulation of Couching Act, Central Provinces and Berar.	1944	II
Regulation of Couching Maharashtra (Extension and Amendment) Act, Central Provinces and Berar.	1963	X
Regulation of Marriage Bureaus and Registration of Marriages Act, Maharashtra.	1999	XX
Regulation of Sugar-cane Price (Supplied to Factories) Act, Maharashtra.	2013	XXXIII
Regularisation of Unauthorised Developments in the City of Ulhasnagar Act, Maharashtra.	2006	IX
Relief Undertaking (Special Provisions) Act, Maharashtra	1958	XCVI
Religious Endownments (Reconstruction on Resettlement Sites) Act, Maharashtra.	1970	XXX
Removal of Disqualifications (of Holders of Office in Co-operative Societies) Act, Maharashtra.	1963	XXIII
Rent Control Act, Maharashtra.	2000	XVIII
Repealing and Amending Act, Bombay.	1949	LIII
Repealing and Amending Act, Bombay (Second).	1951	XXXIX
Repealing and Amending Act, Bombay.	1954	XXI
Repealing and Amending Act, Bombay.	1955	II
Repealing and Amending Act, Bombay (Second).	1955	XXIX
Repealing and Amending Act, Bombay.	1959	LXIII
Repealing and Amending Act, Maharashtra.	1963	XXVI
Repealing and Amending Act, Maharashtra.	1976	XI
Repealing and Amending Act, Maharashtra.	1991	XXII
Repealing and Amending Act, Maharashtra.	2016	XXXI
Repealing and Amending Act, Maharashtra.	2017	XV
Requisitioned Land (Continuance of Power) Act, (M. P.).	1951	IX

Title of the Act	Year	Number of the Act
Restorations of Lands to Scheduled Tribes Act, Maharashtra.	1975	XIV
Restoration of name "Mumbai" for "Bombay" Act, Maharashtra.	1996	XXV
Requisitioning (and Control) of Motor Vehicles Act, Maharashtra.	1966	II
Revenue Patels (Abolition of Office) Act, Maharashtra.	1962	XXXV
Revision of Land Revenue of Mahals (C. P. and Berar) Act.	1947	XXV
Revocation of Land Revenue Exemption Act, (M. P.).	1948	XXXVII
Right to Information (Repeal) Act, Maharashtra.	2006	I
Right to Public Services Act, Maharashtra.	2015	XXXI
${f s}$		
Sales Tax (Amendment and Validating Provisions) Act, Bombay.	1977	XXXVII
Sales of Trees by Occupants belonging to Scheduled Tribes (Regulation) Act, Maharashtra.	1969	XXIII
Salsette Estates (Land Revenue Exemption Abolition) Act.	1951	XLVII
Sandip University Act, Maharashtra.	2015	XXXVIII
Sanjay Ghodawat University Kolhapur Act, Maharashtra.	2017	XL
Satara, Solapur and Southern Maratha Country Laws Act.	1863	III
Scheduled Castes, Scheduled Tribes, De-notified Tribes (Vimukta Jatis), Nomadic Tribes, Other Backward Classes and Special Backward Category (Regulation of Issuance and Verification of) Caste Certificate Act, Maharashtra.	2001	XXIII
Seals Act, Bombay.	1949	XXII
Seals Act (Application in Scheduled Areas) Regulation, Bombay,	1951	Reg. I
Secondary and Higher Secondary Education Boards Act, Maharashtra.	1965	XLI
Secondary School Certificate Examination Act (Application to Scheduled Areas) Regulation, Bombay.	1951	Reg. II
Self-financed Schools (Establishment and Regulation) Act, Maharashtra.	2013	I
Separation of Judicial and Executive Functions Act, Bombay.	1951	XXIII
Separation of Judicial and Executive Functions (Supplementary) Act, Bombay.	1954	VIII
Separation of Judicial and Executive Functions (Extension) and the Code of Criminal Procedure (Provisions for Uniformity) Act, Bombay.	1958	XCVII
Service Inams (Useful to Community) Abolition Act, Maharashtra.	1953	LXX
Settlement of Arrears in Disputes Act, Maharashtra.	2016	XVI
Shetgi Watan Rights (Ratnagiri) Abolition Act, Maharashtra.	1957	II
Shilotri Rights (Kolaba) Abolition Act, Maharashtra.	1955	XLVII

Title of the Act	Year	Number of the Act
Shivraj Fine Art Litho Works (Acquisition and Transfer of Undertaking) Act.	1984	XXXVI
Shops and Establishments(Regulation of Employment and Conditions of Service) Act, Maharashtra.	2017	LXI
Shree Karveer Niwasini Mahalaxmi (Ambabai) Mandir (Kolhapur) Act, Maharashtra.	2018	XXXV
Shree Sai Baba Sansthan Trust (Shirdi) Act, Maharashtra.	2004	XIV
Shri. Shanaishwar Devasthan Trust (Singnapur) Act, Maharashtra.	2018	LVIII
Shree Siddhi Vinayak Ganapati Temple Trust (Prabhadevi) Act.	1980	VI
Sinhastha Fair Pilgrim Tax Act.	1980	II
Slaughter of Animals Act (Hyderabad).	1950AD	VII
Slum Areas (Improvement, Clearance and Redevelopment) Act, Maharashtra.	1971	XXVIII
Special Provision for Payment of Stamp Duty Act, Maharashtra.	1974	XXXIII
Special Provision for payment of Stamp Duty (No. 2) Act, Maharashtra.	1974	LII
Special Suits and Proceedings Validating Act, Bombay.	1951	XV
Special Tribunal (Continuance) Act, Bombay.	1946	XXI
Specified Co-operative Societies (Postponement of Elections due to Scarcity Conditions of the State) Act, Maharashtra.	1972	XLV
Specified Co-operative Societies (Postponement of Elections due to Emergency) Act, Maharashtra.	1975	XLIII
Specified Co-operative Societies (Postponement of Elections due to drought conditions in the State) Act, Maharashtra.	1983	XXVIII
Spicer Adventist University Act, Maharashtra.	2014	XIV
Stamp Act, Maharashtra.	1958	LX
Stamp (Increase of Duties and Amendment) Act, Bombay.	1962	XXXI
State Aid to Industries (C. P. and Berar).	1933	XII
State Aid to Industries, C. P. and Berar and State Aid of (Small Scale and Cottage) Industries, Hyderabad (Partial Repeal) Act.	1959	LXVII
State Aid to Industries Act, Maharashtra.	1960	XVII
State Board of Nursing and Paramedical Education Act, Maharashtra.	2013	XXIII
State Board of Technical Education Act, Maharashtra.	1997	XXXVIII
State Commission for Backward Classes Act, Maharashtra.	2006	XXXIV
State Commission for Safai Karmacharies Act, Maharashtra.	1997	XLIV
State Commission for Women Act.	1993	XV

Title of the Act	Year	Number of the Act
State Council for Occupational Therapy and Physiotherapy Act, Maharashtra.	2004	II
State Council of Examinations Act, Maharashtra.	1999	I
State Enterprises (Restructuring and Other Special Provisions) (Repeal) Act, Maharashtra.	2007	IX
State Election Commissioner (Qualifications and Appointment) Act.	1994	XXII
State Guarantee (Repeal) Act, Bombay.	1965	XXI
State Legislature Proceedings (Protection of Publication) Act, Maharashtra.	1969	XXXII
State Prisoners Regulation Act (Bengal).	1818	III
State Minorities Commission Act, Maharashtra.	2005	XXXI
State Public Services (Reservation for Scheduled Castes, Scheduled Tribes, De-notified Tribes (Vimukta Jatis), Nomadic Tribes, Special Backward Category and Other Backward Classes) Act, Maharashtra.	2004	VIII
State Reservation (of seats for admission in educational institutions in the State and for appointments in the public services and posts under the State) for Socially and Educationally Backward Classes (SEBC) Act, Maharashtra.	2018	LXII
State Reserve Police Force Act, Maharashtra.	1951	XXXVIII
State Scarcity Relief Fund (Repeal) Act, Bombay.	2008	V
State Security Corporation Act, Maharashtra.	2010	VI
State Tax on Professions, Trades, Callings and Employments Act, Maharashtra.	1975	XVI
Statutory Corporations (Regional Reorganisation) Act, Bombay.	1960	XI
Statutory Funds Act, Maharashtra.	1959	LXXI
Steam Vessels Act, Maharashtra.	1864	II
Surgeon General With Government etc. (Change in Designations) Act.	1973	III
Supply of Forest-Produce by Government (Revision of Agreement Act) Maharashtra.	1983	XVI
Suraksha Dal Act, Maharashtra.	1987	III
Symbiosis Skills and Open University Act, Maharashtra.	2017	XXXVII
T		
Tapi Irrigation Development Corporation Act, Maharashtra.	1998	IV
Tariff (Bombay Application) Regulation Indian.	1943	Reg. I

Title of the Act	Year	Number of the Act
Taxation Laws Amendment (Inapplicability of Limitation) Act, Maharashtra.	1976	XXIV
Taxation Laws Offences (Extension of Period of Limitation) Act, Maharashtra.	1977	XLIV
Taxation Laws Offences (Extension of Period of Limitation) Act, Maharashtra.	1982	XXII
Tax on Buildings (with larger Residential Premises) (Re-enacted) Act, Maharashtra.	1979	XXIX
Tax on Sale of Electricity Act, Maharashtra.	1963	XXI
Temporary Increase in Entertainment Duty and Education Cess Act, Maharashtra.	1972	VII
Temporary Increase in Taxes on Motor Vehicles and Passengers Act, Maharashtra.	1972	XI
Tenancy and Agricultural Lands Act, Maharashtra.	1948	LXVII
Tenancy and Agricultural Lands (Vidarbha Region) Act, Maharashtra.	1958	XCIX
Tenancy and Agricultural Laws (Amendment) Act, Hyderabad.	1959	XXXII
Tenancy and Agricultural Lands (Amendment) Act, Hyderabad.	1960	XXVII
Tenancy and Agricultural Lands (Re-enactment) Validation and further (Amendment) Act, Hyderabad.	1961	XLV
Tenancy Law and Maharashtra Land Revenue Code (Amendment and Validation of Appointments of Certain Officers and Proceedings) Act, Maharashtra.	1982	V
Textile Companies (Acquisition and transfer of Undertakings) Act, Maharashtra.	1982	XXXIII
Tobacco Laws (Repeal) Act, Bombay.	1959	LXXIII
Town Planning Schemes [Bombay City Nos. II, III and IV (Mahim Area)], Validation Act, Bombay.	1957	III
Town Planning Act (C. P. and Berar).	1948	LXVII
Tribal Areas Regulation (Hyderabad).	$1359\mathrm{F}$	Reg. Ill
Tribal Economic Condition (Improvement) Act.	1977	V
Truck Terminal (Regulation of Location) Act, Maharashtra.	1997	XXXI
Tuberculosis Sanatorium (Regulation of Buildings) Act (C.P. and Berar).	1948	XIII
Tuljapur Development Authority Act.	2009	III
${f U}$		
Unaided Private Professional Educational Institutions (Regulation of Admissions and Fees) Act, Maharashtra.	2015	XXVIII
Unauthorised Institutions and Unauthorised Courses of Study in Agriculture, Animal and Fishery Sciences, Health Sciences, Higher, Technical and Vocational Education (Prohibition) Act, Maharashtra.	2013	XX

Title of the Act	Year	Number of the Act
Unclaimed Property Act (Hyderabad).	1333 F	I
Unemployment Allowance Payment to Workmen in Factories (for Temporary Period) Act, Maharashtra.	1976	XIV
Industrial Disputes and the Maharashtra Un-employment Allowance Payment to workmen in factories (for Temporary period) (Amendment) Act.	1981	XXII
Underground Pipelines and Underground Ducts (Acquisition of Right of user in Land) Act, Maharashtra.	2018	XLIX
Urban Areas Preservation of Trees Act, Maharashtra.	1975	XLIV
University of Health Sciences Act, Maharashtra.	1999	X
Public Universities Act, Maharashtra.	2017	VI
Universities (Temporary Postponement of Elector of Members of University Authorities and Other Bodies) Act, Maharashtra.	2015	XXIX
Universities (Temporary Postponement of Elections of Members of University Authorities and Other Bodies) Act, Maharashtra.	2000	XXIV
University (Temporary Postponement of Election of University Representatives) Act, Bombay.	1989	XL
Universities (Temporary Postponement of Elections of members of University authorities and other bodies) Act, Maharashtra.	1993	XXVII
${f v}$		
Vacant Lands (Prohibition of Unauthorised Occupation and Summery Eviction) Act, Maharashtra.	1975	LXVI
Vacant Lands (Further Interim Protection to Occupiers from Eviction and Recovery of Arrears of Rent) Act.	1980	XVI
Value Added Tax Act, Maharashtra.	2005	IX
Veterinary Practitioners Act, Maharashtra.	1971	XLIX
Vetti and Beggar (Prohibition) Regulation (Hyderabad).	1358 F	XLIII
Vexatious Litigation (Prevention) Act, Maharashtra.	1971	XLVIII
Vidarbha Irrigation Development Corporation Act.	1997	XXVI
Vijaybhoomi University, Raigad Act, Maharashtra.	2018	LXXVII
Village Panchayats Act, Maharashtra.	1959	III
Village Panchayats (Postponement of Elections during the Emergency) Act, Maharashtra.	1975	XLVIII
Village Panchayats (Postponement of Elections due to ensuing General Elections to the State Legislative Assembly, Zilla Parishads and Panchayat Samitis) Act, Maharashtra.	1977	LI
Village Panchayats (Temporary Postponement of Elections due to Election of Lok Sabha) Act.	1980	III

Title of the Act	Year	Number of the Act
Village Panchayats (Temporary further Postponement of Elections due to Elections to Maharashtra Legislative Assembly) Act, Maharashtra.	1980	XIV
Village Panchayats (Temporary Postponement of Elections due to Preparation of Revised Assembly Roll) Act, Maharashtra.	1983	XLVI
Village Panchayats (Temporary Postponement of Certain Elections due to drought Conditions in the State) Act, Maharashtra.	1987	IX
Village Panchayats (Temporary Postponement of Certain Elections) Act, Maharashtra.	1989	XXXVIII
Vishwakarma University Act, Maharashtra.	2017	XXXVIII
Village Police Act, Maharashtra.	1967	XLVI
W		
Warehouses Act, Maharashtra.	1960	V
Water Conservation Corporation Act, Maharashtra.	2001	III
Water Resources Regulatory Authority Act, Maharashtra.	2005	XVIII
Weekly Holiday (Substitution of Sunday for Friday) Act, (Hyderabad).	1950A.D.	VIII
West Khandesh Mehwassi Estates Regulation.	1949	Reg. I
West Khandesh Mehwassi Estates (Proprietary Rights Abolition, etc.) Regulation.	1962	Reg. I
Wills and Intestacy Regulation Section 4, 5, 6 and 7 (Bengal) (M.P.).	1799	V
Witness Protection and Security Act, Maharashtra.	2018	IX
Workmen's Minimum House Rent Allowances Act, Maharashtra. ${\bf Y}$	1988	XXIII
Yeshwantrao Chavan Maharashtra Open University Act.	1985	XX
${f z}$		
Zilla Parishads and Panchayat Samitis Act, Maharashtra.	1961	V
Zilla Parishads and Panchayat Samitis (Postponement of Elections to Zilla Parishads and Panchayat Samitis and (Amendment) Act.	1979	X
Zilla Parishads and Panchayat Samitis Temporary Postponements of Elections (of the President, Vice President and Chairman of Subjects Committees of certain Zilla Parishads and Chairman and Deputy Chairman of certain Panchayat Samitis due to ensuing general elections to the State Legislative Assembly and certain other elections) Act, Maharashtra.	2005	VII

PART II

Local Acts of the Governor-General in Council in force in the State of

Maharashtra including (Unrepealed) Regulations

Title of the Act	Year	Number of the Act
A		
Acknowledgment of Debts, Interest, Mortgages—Regulation.	1827	Reg. V
Administration of Estates—Regulation.	1827	Reg. VIII
В		
Baronetcy Act, Sir Dinshaw Maneckjee Petit.	1893	VI
Baronetcy Act, Sir Jamsetjee-Jeejeebhoy.	1915	X
${f C}$		
City of Bombay Municipal (Supplementary) Act.	1888	XII
Civil Courts Act, Maharashtra.	1869	XIV
Civil Courts (Enhancement of Pecuniary Jurisdication and Amendment) Act (Maharashtra).	1977	XLVI
Civil Courts (Law to be observed) Regulation.	1827	Bom. Reg. IV
D		
Dekkhan and Khandesh (Pune, Ahmednagar and Khandesh Districts)—Regulation.	1827	Bom. Reg. XXIX
${f E}$		
Easements (Extending Act 5 of 1882).	1891	VIII
H		
High Court (Letters Patent) Act, Bombay.	1866	XXIII
High Court (Letters Patent Amendment) Act.	1948	XLI
${f L}$		
Land Suits-Regulation.	1831	Bom. Reg. I
M		
Military Authority (Assistance to Marching Troops) Regulation.	1827	Bom. Reg. XXII
Municipal Debentures Act, Bombay.	1876	XV
Municipal (Supplementary) Act, City of Bombay.	1888	XII
${f R}$		
Rent-Free Estates Act, Bombay.	1852	XI
Revenue Jurisdiction Act, Maharashtra.	1876	X

Title of the Act	Year	Number of the Act
S		
Seals Act (Application to Scheduled Areas) Regulation (Bombay).	1951	I
Secondary School Certificates Examination Act (Application to Scheduled Areas) Regulation (Bombay).	1951	II
Shore Nuisances (Bombay and Kolaba) Act.	1853	XI
T		
Tariff (Bombay Application) Regulation, Indian.	1943	I
Tribal Areas Regulation.	1359F	III
V		
Vetti and Begar (Prohibition) Regulation (Hyderabad).	1358F	XLIII
\mathbf{w}		
West Khandesh Mehwassi Estates Regulation.	1949	I
West Khandesh Mehwassi Estates (Proprietory Rights Abolition, etc.) Regulation.	1962	I

PART III Bombay and Maharashtra Acts Amending Unrepealed Central Acts in their application to the State of Maharashtra

STATE ACTS AND REGULATIONS

Title of the Act	Year	Number of the Act
A		
Administrator-General and Official Trustees (Maharashtra Amendment) Act, 1982.	1983	XIV
Apprentices (Maharashtra Amendment) Act.	2018	XVII
${f C}$		
Cattle-trespass (Bombay Amendment) Act.	1931	V
Cattle-trespass and Bombay District Police (Amendment) Act.	1950	XXXIV
Cattle-trespass and (Bombay Extension and Amendment) Act.	1959	XIII
Cigarettes and Other Tobacco Products (Prohibition of Advertisement and Regulation of Trade and Commerce, Production, Supply and Distribution) (Maharashtra Amendment) Act.	2018	LX
Civil Courts (Enhancement of Pecuniary Jurisdiction and Amendment) Act.	1977	XLVI
Code of Civil Procedure (Bombay Amendment) Act.	1948	LX
Code of Civil Procedure (Extension of Hyderabad Amendment) Act.	1965	VI
Code of Civil Procedure (Maharashtra Amendment) Act.	1970	XXV
Code of Civil Procedure (Maharashtra Amendment) Act.	1977	LXV
Code of Civil Procedure (Maharashtra Amendment) Act.	2018	LXI
Code of Civil Procedure (Maharashtra Amendment) (Amendment) Act.	2018	LXXII
Code of Criminal Procedure (Maharashtra Amendment) Act.	1976	XXIII
Code of Criminal Procedure (Maharashtra Amendment) Act, 1977.	1978	I
Code of Criminal Procedure (Maharashtra Amendment) Act, 1981.	1981	XXXIV
Code of Criminal Procedure (Maharashtra Second Amendment) Act, 1981.	1981	LX
Code of Criminal Procedure (Maharashtra Amendment) Act, 1993.	1993	XXIV
Code of Criminal Procedure (Maharashtra Amendment) Act, 1998.	1999	XXI
Code of Criminal Procedure (Maharashtra Amendment) Act, 2004.	2005	VIII

Title of the Act	Year	Number of the Act
Code of Criminal Procedure (Maharashtra Amendment) Act, 2007.	2007	XXVII
Code of Criminal Procedure (Maharashtra Amendment) Act, 2006.	2014	XXXIII
Code of Criminal Procedure (Maharashtra Amendment) Act, 2015.	2016	XXXIII
Indian Penal Code and Code of Criminal Procedure (Maharashtra Amendment) Act, 2017.	2017	XL
Commercial Documents Evidence (Maharashtra Extension) Act.	1960	XX
Contract Labour (Regulation and Abolition) (Maharashtra Amendment) Act.	2017	II
Cotton Ginning and Pressing Factories (Bombay Amendment) Act.	1936	IV
Cotton Ginning and Pressing Factories (Bombay Amendment) Act.	1938	XX
Cotton Ginning and Pressing Factories (Bombay Amendment) Act.	1956	XXVII
Cotton Ginning and Pressing Factories (Bombay Amendment) Act.	1958	XX
Cotton Ginning and Pressing Factories (Maharashtra Provision for Uniform Application and Amendment) Act.	1962	I
Cotton Ginning and Pressing Factories (Maharashtra Amendment) Act.	1974	II
Criminal Law Amendment (Bombay Amendment and Extension) Act.	1959	XXIV
Criminal Law Amendment (Maharashtra Amendment) Act. D	1975	XXXVIII
Drugs and Cosmetics (Maharashtra Amendment) Act.	1989	XXXIE
E	2000	
Electricity (Maharashtra Amendment) Act.	2005	XXXVI
Epidemic Diseases (Bombay Amendment) Act.	1953	XII
Evacuee Interest (Separation) Validating and Supplementary Act, Bombay.	1953	XXIX
Evacuee Interest (Separation) Supplementary Act Maharashtra.	1960	XXV
Essential Commodities (Maharashtra Amendment) Act, 1975.	1976	I
${f F}$		
Factories (Maharashtra Amendment) Act.	2006	XXVIII
Factories (Maharashtra Amendment) Act.	2015	XL

Title of the Act	Year	Number of the Act
Forest (Bombay Amendment) Act, Indian.	1948	LXII
Forest (Bombay Amendment) Act, Indian.	1953	XXV
Forest (Bombay Amendment) Act, Indian.	1955	XXIV
Forest (Bombay Amendment) Act, Indian.	1956	X
Forest (Bombay Second Amendment) Act, Indian.	1956	XVII
Forest (Bombay Amendment) Act, Indian.	1957	XXVI
Forest (Maharashtra Unification and Amendment) Act, Indian.	1961	VI
Forest (Maharashtra Amendment) Act.	1965	XV
Forest (Maharashtra Amendment) Act.	1968	XXVII
Forest (Maharashtra Second Amendment) Act, 1984, Indian.	1984	XXIII
Forest (Maharashtra Amendment) Act, 1984, Indian.	1985	VII
Forest (Maharashtra Amendment) Act, Indian.	2015	XXI
${f G}$		
Glanders and Farcy and Dourine Acts (Extension) Act.	1960	XXXIV
Н		
High Court (Hearing of Writ Petitions by Division Bench and Abolition of Letters Patent Appeals) Act, Maharashtra.	1986	XVII
Hindu Succession (Maharashtra Amendment) Act.	1994	XL
I		
Indian Penal Code and Code of Criminal Procedure (Maharashtra Amendment) Act, 2017.	2017	XL
Indian Partnership (Maharashtra Amendment) Act.	2018	XVI
Industrial Disputes (Bombay Provision for Uniformity) Act.	1959	XVI
Industrial Disputes (Maharashtra Amendment) Act.	1963	II
Industrial Disputes (Maharashtra Amendment) Act.	1974	LVI
Industrial Disputes (Maharashtra Amendment) Act.	1976	XXII
Industrial Disputes (Maharashtra Amendment) Act, 1981.	1982	III
Industrial Disputes (Appellate Tribunal) (Bombay Amendment) Act.	1953	XVI
Industrial Disputes (Appellate Tribunal) (Bombay Amendment) Act.	1954	LXV
Industrial Disputes and Maharashtra Unemployment Allowance Payment to Workmen in Factories for Temporary Period (Amendment) Act.	1981	XXII
Industrial Employment (Standing Orders) (Bombay Amendment) Act.	1955	LIII
Industrial Employment (Standing Orders) (Bombay Amendment) Act.	1956	XXXVI

Title of the Act	Year	Number of the Act
Industrial Employment (Standing Orders) (Bombay Amendment) Act.	1958	XXI
Industrial Employment (Standing Orders) (Maharashtra Amendment) Act.	1974	LIV
Interest (Extension) Act.	1960	XXIII
J		
Judicial Officer's Protection (Extension to Hyderabad and Saurashtra Areas of Bombay State) Act.	1958	LXXVIII
K		
Kazis (Maharashtra Amendment) Act, 1978.	1978	XXI
${f L}$		
Land Acquisition (Maharashtra Amendment) and Maharashtra Regional and Town Planning (Amendment) Act.	1973	XLII
Land Improvement Loans and Agriculturists Loans (Extension and Amendment) Act.	1958	XXVII
Legal Practitioners (Bombay Amendment) Act.	1948	XLV
(Letters Patent) Act, Bombay High Court.	1866	XXIII
Letters Patents (Amendment) Act, Bombay High Court.	1948	XLI
Local Authorities Loans (Bombay Amendment) Act.	1942	II
M		
Medical Degrees (Maharashtra Extension and Provision for Uniformity) Act, Indian.	1961	XLIV
Minimum Wages (Maharashtra Amendment) Act.	1961	X
Minimum Wages (Maharashtra Amendment) Act.	1963	III
Minimum Wages (Maharashtra Amendment) Act, 1975.	1976	XXV
Motor Vehicles (Maharashtra Amendment) Act.	1972	X
Motor Vehicles (Maharashtra Second Amendment) Act.	1973	VI
Motor Vehicles (Maharashtra Amendment) Act.	1975	XIX
Motor Vehicles (Maharashtra Amendment) Act.	1976	XLVII
0		
Orphanages and other Charitable Homes (Supervision and Control), the Persons with Disabilities (Equal Opportunities, Protection of Rights and Full Participation) and the Building and other Construction Workers (Regulation of Employment and Conditions of Service) (Maharashtra Amendment) Act.	2016	IV

Title of the Act	Year	Number of the Act
P		
Partnership (Maharashtra Amendment) Act, 1984, Indian.	1984	XXIX
Payment of Bonus (Maharashtra Amendment) Act.	2011	III
Payment of Wages (Bombay Amendment) Act.	1953	LXII
Payment of Wages (Bombay Amendment) Act.	1954	LXX
Payment of Wages (Bombay Amendment) Act.	1955	XLVIII
Payment of Wages (Bombay Amendment) Act.	1956	XXXVII
Payment of Wages (Bombay Amendment and Validation) Act.	1960	VIII
Payment of Wages (Maharashtra Provision for Uniform Application and Amendment) Act.	1961	XIII
Payment of Wages (Maharashtra Amendment) Act.	1961	XLII
Payment of Wages (Maharashtra Amendment) Act.	2011	III
Payment of Wages and minimum wages (Maharashtra Amendment) Act.	2010	XVI
Police (Incitement to Disaffection) (Bombay Extension and Amendment) Act.	1958	LXXVII
Police (Incitement to Disaffection) (Maharashtra Amendment) Act, 1983.	1983	XXIII
Ports (Maharashtra Amendment) Act, Indian.	1963	IV
Ports (Maharashtra Amendment) Act, Indian.	1994	LV
Ports (Amendment) Act, Bombay Landing and Wharfage Fees and Indian.	1973	VIII
Post Office (Bombay Amendment) Regulation, Indian.	1942	Reg-I
Presidency Small Cause Courts (Bombay Amendment) Act.	1933	V
Presidency Small Cause Courts (Bombay amendment) Act.	1948	XLIV
Presidency Small Cause Courts (Bombay Amendment) Act.	1952	XVII
Presidency Small Cause Courts (Bombay Amendment) Act.	1953	LIX
Presidency Small Cause Courts (Bombay Amendment) Act.	1955	XLIV
Presidency Small Cause Courts (Bombay Amendment) Act.	1959	XI
Presidency Small Cause Courts (Maharashtra Amendment) Act.	1961	XXXV
Presidency Small Cause Courts (Maharashtra Amendment) Act.	1963	XLI
Presidency Small Cause Courts (Maharashtra Amendment) Act.	1966	XV
Presidency Small Cause Courts (Maharashtra Amendment) Act, 1975.	1976	XIX
Presidency Small Cause Courts (Maharashtra Amendment) Act.	1977	LIX
Presidency Small Cause Courts (Maharashtra Amendment) Act.	1987	XXVI
Press and Registration of Books (Bombay Amendment) Act.	1948	LXI

Title of the Act	Year	Number of the Act
Press and Registration of Books (Bombay Amendment) Act.	1951	VI
Prevention of Cruelty to Animals (Maharashtra Amendment) Act.	2017	XLV
Prisoners (Bombay Amendment) Act.	1947	XVII
Prisoners (Bombay Amendment) Act.	1949	VII
Prisoners (Bombay Extension and Amendment) Act.	1959	XV
Prisoners (Bombay Amendment) Act, Identification of—	1948	XV
Prisoners (Bombay Amendment) Act, Identification of—	1953	LVIII
Prisoners (Maharashtra Provision for Uniform Application and Amendment) Act, Identification of—	1970	XXXV
Prisons (Bombay Amendment) Act.	1953	XXVII
Prisons and Civil Jails (Bombay Extension, Unification and Amendment) Act.	1959	XXIII
Prisons (Bombay Amendment) Act.	1959	XLV
Probation of Offenders (Maharashtra Amendment) Act.	1969	XXXI
${f R}$		
Right to Fair Compensation and Transparency in Land Acquisition, Rehabilitation and Resettlement (Maharashtra Amendment) Act.	2018	XXXVII
${f T}$		
Tariff (Bombay Application) Regulation, 1943 Indian.	1943	Bom. Reg. I
Trade Unions (Bombay Amendment) Act, Indian.	1956	XXXII
Trade Unions and Bombay Industrial Relations (Amendment) Act.	1968	III
Tramways (Bombay Amendment) Act, Indian.	1948	${f L}$
Transfer of Property and the Indian Registration (Bombay Amendment) Act.	1939	XIV
Transfer of Property (Bombay provision for Uniformity and	1959	LVII
Amendment) Act.		
Treasur-trove (Extension to the Hyderabad and Saurashtra Areas of Bombay State) Act, Indian.	1958	XXXIII

PART IV

SUBJECTWISE INDEX

To State Acts and Regulations in Force in the State of Maharashtra

ABKARI—

Bombay Abkari Act, 1878 (Bom. V of 1878).

Bombay Abkari (Amendment) Act, 1912 (Bom. XII of 1912).

ABOLITION OF-

Cash Grant—See "Land Tenure".

Inams—See "Land Tenure".

Jagirs—See "Land Tenure".

Proprietory Rights—See "Land Tenure".

Khoti—See "Land Tenure".

Land Tenure—See "Land Tenure".

Miscellaneous Alienation—See "Land Tenure".

Watans—See "Land Tenure".

Proprietory Rights in Mines and Minerals—See "Mines and Minerals".

Proprietory Rights in Estates, Mahals, etc., See "Land Tenure".

Whipping—See "Criminal Law and Procedure".

Salami Tenure—See "Land Tenure".

Ankadia Tenure—See "Land Tenure".

Service Inam—See "Land Tenure".

Shilotri Rights—See "Land Tenure".

Udhad and Ugadia Tenure—See "Land Tenure".

ACQUISITION AND REQUISITION—

Requisitioned Land (Continuance of Power) Act, (M.P.) 1951 (IX of 1951).

Land Acquisition (Bombay Amendment) Act, 1950 (Bom. XXVII of 1950).

Land Acquisition (Bombay Amendment) Act, 1953 (Bom. XXXV of 1953).

Land Acquisition (Bombay Amendment) Act, 1957 (Bom. XII of 1958).

Land Acquisition (Maharashtra Extension and Amendment) Act, 1964 (Mah. XXXVIII of 1964).

Land Acquisition (Maharashtra Amendment and Validation of Certain Proceedings for Acquisition of Lands) Act, 1965 (Mah. XXIV of 1965).

Maharashtra Requisitioning (and Control) of Motor Vehicles Act, 1965 (Mah. II of 1966).

Land Acquisition (Maharashtra Amendment) Act, 1972 (Mah. XXXIX of 1972).

Land Acquisition (Maharashtra Amendment) and Maharashtra Regional and Town Planning (Amendment) Act, 1973 (Mah. XLII of 1973).

Bombay Land Acquisition Officers Proceedings Validation Act, 1949 (Bom. XXXV of 1949).

Central India Spinning, Weaving and Manufacturing Company Limited, the Empress Mills, Nagpur (Acquisition and Transfer of Undertaking) Act, 1986 (XLVI of 1986).

Maharashtra Textile Companies (Acquisition and Transfer of Undertaking) Act, 1982 (Mah. XXXIII of 1982).

Pulgaon Cotton Mills Limited (Acquisition of Shares) Act, 1982 (Mah. XXXIV of 1982).

Shivraj Fine Art Litho Works (Acquisition and Transfer of Undertaking) Act, 1984 (Mah. XXXVI of 1984).

ADMINISTRATION OF ESTATES—

Bombay Regulation VIII of 1827.

Agricultural Debtors Relief (Mehwassi Estates and Akrani Mahal) (Amendment) Regulation, Bombay 1952 (Bom. Reg. II of 1952).

Attached Estates Management Regulation (Bengal), 1827 (Reg. V of 1827).

West Khandesh Mehawassi Estates (Proprietory Rights Abolition) Regulation, 1961 (Mah. Reg. I of 1962).

Salsette Estate (Land Revenue Exemption Abolition) Act, 1951 (Bom. XLVII of 1951).

ADVOCATE—

Maharashtra Advocates Welfare Fund Act, 1981 (Mah. LXI of 1981).

AGRICULTURE—

Alluvial and Dilluvial Regulation, 1825 (Bengal) (XI of 1825).

Attached Estate Management Regulations, 1827 (Bengal) (V of 1827).

Land Alienation Act, 1916 (C.P. and Berar) (II of 1916).

Debt Conciliation Act, 1916 (C.P. & Berar) (II of 1916).

Bombay Fodder and Grain Control Act, 1939 (Bom. XXVI of 1939).

Bombay Land Improvement Schemes Act, 1942 (Bom. XXVIII of 1942).

Bombay Agricultural Debtors Relief Act, 1947 (Bom. XXVIII of 1947).

Bombay Agricultural Pests and Diseases Act, 1947 (Bom. XLIII of 1947).

Bombay Prevention of Fragmentation and Consolidation of Floldings Act, 1947 (Bom. LXII of 1947).

Bombay Non-Agriculturists' Loans Act, 1928 (Bom. Ill of 1928).

Bombay Tenancy and Agricultural Lands Act, 1948 (Bom. LXVII of 1948).

Reclamation of Lands (Eradication of Kans) Act, 1948 (C.P. and Berar) (LXVII of 1948). Agricultural Improvement Fund Act (Hyderabad) 1952 (XLV of 1952).

Bhudan Yagna Act, 1953 (M.P.) (XV of 1953).

Agricultural Income-tax (Validation of Notices) Act, 1954 (Hyderabad) (II of 1954).

Agricultural Debtors Relief Act, 1956 (Hyderabad) (XVI of 1956).

Maharashtra Tenancy and Agricultural Lands (Vidarbha Region) Act (XCIX of 1958).

Land Improvement Loans (Extension and Amendment) Act, 1958 (Bom. XXVII of 1958).

Hyderabad Tenancy and Agricultural Lands (Amendment) Act, 1959 (Bom. XXXII of 1959).

Hyderabad Tenancy and Agricultural Lands (Amendment) Act, 1960 (Mah. XXVIII of 1960).

Hyderabad Tenancy and Agricultural Lands (Re-enactment, Validation and further Amendment) Act, 1961 (Mah. XLV of 1961).

Maharashtra Agricultural Produce Marketing (Regulation and Development) Act, 1963 (Mah. XX of 1964).

Agriculturists' Loans (Maharashtra Amendment) Act, 1965 (Mah. XXXVIII of 1965). Maharashtra Provision of Facilities for Agricultural Credit by Banks Act, 1974 (Mah. V of 1975).

Maharashtra Restoration of Lands to Scheduled Tribes Act, 1974 (Mah. XIV of 1975). Maharashtra Agricultural Produce Marketing (Regulation) (Postponement of Elections during the Emergency) Act, 1975 (Mah. LXVII of 1975).

Maharashtra Debt Relief Act, 1975 (Mah. Ill of 1976).

Protection of Debtors (C.P. and Berar) Act, 1937 (IV of 1937).

ALLOWANCES—

See "Salaries, Allowances and Pension".

ANATOMY—

Bombay Anatomy Act, 1949 (Bom. XI of 1949).

ANIMALS AND BIRDS—

Prevention of Cruelty to Animals Act, 1313 F (Hyderabad) (1 of 1313 F).

Bombay Livestock Improvement Act, 1933 (Bom. XXII of 1933).

Bombay Diseases of Animals Act, 1948 (Bom. LIX of 1948).

Animal Preservation Act, 1977 (Mah. IX of 1977).

AERIAL ROPEWAYS—

See "Communications".

AUDIT LOCAL FUND—

See "Local Authorities".

BARONETCY—

See "Private Property-Administration of".

BEGGARS—

Vetti and Beggar (Prohibition) Regulation (Hyderabad), 1358 F (XLIII of 1358 F).

Prevention of Begging Act, 1959 (Bom. X of 1960).

BOARD—

Maharashtra Maritime Board Act, 1996 (Mah. XV of 1997).

See "Education".

BOOKS-

See "Press, Printing and Publications".

BORSTAL SCHOOLS—

See "Prisons, Prisoners and Reformatories".

BUILDING—

City of Bombay (Building Works Restriction) Act, 1949 (Bom. XLIV of 1949).

Bombay Building (Control on Erection) Regulation, 1950 (Bom. Reg. I of 1950).

Poona Mutha River Flood Limits (Prohibition of Building and Provision for Alternative Building Sites) (Repeal) and Indemnity Act, 1965 (Mah. X of 1966).

Bombay Queen Victoria Statue Site (And adjoining land utilisation for construction of Satelite, Telecommunications Exchanges of the Overseas Communications Services) Act, 1968 (Mah. V of 1968).

Maharashtra Religious Endowments (Re-construction on Re-settlement Sites) Act, 1970 (Mah. XXX of 1970).

Bombay Building (Control on Erection, Re-erection and Conversion) (Repeal) Act, 1971 (Mah. XLI of 1971).

Maharashtra Apartment Ownership Act, 1971 (Mah. XV of 1971).

Maharashtra Ownership Flats (Regulation of the promotion of construction, sale, management and transfer) Act, 1963 (Mah. XLV of 1963).

Maharashtra Regularisation of Unauthorised Developments in the City of Ulhasnagar Act, 2006 (Mah. IX of 2006).

CASH GRANTS—

See "Land Tenure".

CATTLE—

Cattle Trespass (Bombay Amendment) Act, 1931 (Bom. V of 1931).

Cattle Trespass and Bombay District Police (Amendment) Act, 1950 (Bom. XXXIV of 1950).

Bombay Essential Commodities and Cattle (Control) Act, 1958 (Bom. LXII of 1958).

Cattle Trespass (Bombay Extension and Amendment) Act, 1959 (Bom. XIII of 1959).

Keeping and Movement of Cattle in Urban Areas (Control) Act, 1976, Maharashtra (XLV of 1976).

CENSUS EXPENSES CONTRIBUTION—

See "Local Authorities".

CESS—

Maharashtra Education and Employment Guarantee (Cess) Act, 1962 (Mah. XXVII of 1962).

CHARGED EXPENDITURE—

Bombay Charged Expenditure Act, 1957 (Bom. V of 1957).

Maharashtra Fiscal Responsibility and Budgetary Management Act, 2005 (Mah. XVI of 2005).

CHILDREN—

Children Protection Act, 1343 F (Hyderabad) (IX of 1343 F).

Juvenile Smoking Act, 1929 (C.P. & Berar) (VIII of 1929).

CINEMAS—

Maharashtra Cinemas (Regulation) Act (XI of 1953).

CIVIL PROCEDURE AND PRACTICE—

Bombay Lands Suits Regulation, 1831 (I of 1831).

Bombay Revenue Jurisdiction Act, 1876 (X of 1876).

Bombay Adjudication Proceedings (Transfer and Continuance) Act, 1947 (Bom. XXXIV of 1947).

Bombay Special Suits and Proceedings Validating Act, 1951 (Bom. XV of 1951).

Akrani Mahal (Application of Laws) Regulation, 1952 (Bombay Reg. I of 1952).

Bombay Judicial Proceedings (Regulation of Reports) Act, 1955 (Bom. XVIII of 1955).

Bombay Court-Fees Act, 1959 (Bom. XXXVI of 1959).

Bombay Execution of Decrees (Temporary Postponement) Act, 1959 (Bom. LXX of 1959).

Suits Valuation (Maharashtra Extension and Amendment) Act, 1960 (Mah. IV of 1960).

Commercial Documents Evidence (Maharashtra Extension) Act, 1960 (Mah. XX of 1960).

Suits Valuation and Bombay Court-Fees (Amendment) and Bombay City Civil Court Decrees and Orders (Validating) Act, 1970 (Mah. IX of 1970).

Maharashtra Vexatious Litigation (Prevention) Act, 1971 (Mah. XLVIII of 1971).

Special Provision for Payment of Court-Fees Act, 1976 (Mah. XXXVII of 1976).

CODE OF CIVIL PROCEDURE—

Code of Civil Procedure (Bombay Amendment) Act, 1948 (Bom. LX of 1948).

Code of Civil Procedure (Extension of Hyderabad Amendment) Act, 1964 (Mah. VI of 1965).

Code of Civil Procedure (Maharashtra Amendment) Act, 1970 (Mah. XXV of 1970).

Code of Civil Procedure (Maharashtra Amendment) Act, 1976 (Mah. LXV of 1977).

The Code of Civil Procedure (Maharashtra Amendment) Ad. 2018 (Mah. LIX of 2018)

Special Provision for Payment of Court-Fees Act, 1976 (Mah. XXXII of 1976).

CODE OF CRIMINAL PROCEDURE-See "Criminal Law and Procedure".

COMMISSION—

Maharashtra State Commission for Women Act, 1993 (Mah. XV of 1993).

COMMISSIONERS—

Bombay Commissioners of Divisions Act, 1957 (Bom. VIII of 1958).

Bombay State Commissioners of Police Act, 1959 (Bom. LVI of 1959).

COMMUNICATIONS—

Highways Act, 1936 (C.P. & Berar) (XXXIV of 1936).

Bombay Aerial Ropeways Act, 1955 (Bom. III of 1956).

Maharashtra Highways Act, 1955 (Bom. LV of 1955).

CONSOLIDATION OF HOLDINGS—

See "Agriculture".

CONTAGIOUS DISEASES—

See "Animals and Birds".

CONTINGENCY FUND—

Bombay Contingency Fund (Temporary Amendment) Act, 1978 (Mah. VI of 1978).

Maharashtra Contingency Fund Act, 1956 (Bom. XLVI of 1956).

CONTRACTS—

Bhagelas Contracts Act, 1353 F (Hyderabad) (IX of 1353 F).

Act for Avoiding Wages (Amendment) Act, 1865 (Bom. III of 1865).

Bombay Cotton Contracts Control (War Provisions) Repeal Act, 1922 (Bom. III of 1922).

Bombay Cotton Contracts Act, 1932 (Bom. IV of 1932).

Bombay Forward Contracts Control Act, 1947 (Bom. LXIV of 1947).

CO-OPERATIVE SOCIETIES—

See "Societies".

CORNEAL GRAFTING—

See "Medical Profession".

CORONERS—

Coroners (Maharashtra Repeal) Act, 1999 (Mah. VIII of 2000).

CORPORATIONS—

Maharashtra Non-Trading Corporations Act (XXVI of 1959).

Statutory Corporations (Regional Reorganisation) Act, 1960 (Bom. XXI of 1960).

Also see "Local Authorities".

Maharashtra Godawari Marathwada Irrigation Development Corporation Act, 1998 (Mah. XXIII of 1998).

Konkan Irrigation Development Corporation Act, 1997 (Mah. III of 1998).

Maharashtra Tapi Irrigation Development Corporation Act, 1997 (Mah. IV of 1998).

Vidarbha Irrigation Development Corporation Act, 1997 (Mah. XXVI of 1997).

COTTON—

Cotton Contracts—See "Contracts".

Cotton Ginning and Pressing Factories (Bombay Amendment) Act, 1936 (Bom. IV of 1936) (Principal Act Repealed by Central Act 2 of 1999).

Cotton Ginning and Pressing Factories (Bombay Amendment) Act, 1938 (Bom. XX of 1938) (Principal Act Repealed by Central Act 2 of 1999).

Bombay Cotton Control Act, 1942 (Bom. XXX of 1942).

Bombay Cotton (Statistics) Act, 1946 (Bom. XXVII of 1946).

Cotton Ginning and Pressing Factories (Bombay Amendment) Act, 1956 (Bom. XXVII of 1957) (Principal Act Repealed by Central Act 2 of 1999).

Cotton Ginning and Pressing Factories (Bombay Amendment) Act, 1958 (Bom. XX of 1958) (Principal Act Repealed by Central Act 2 of 1999).

Cotton Ginning and Pressing Factories (Maharashtra Provisions for Uniform Application and Amendment) Act, 1961 (Mah. I of 1962) (Principal Act Repealed by Central Act 2 of 1999).

Cotton Ginning and Pressing Factories (Maharashtra Amendment) Act, 1973 (Mah. II of 1974) (Principal Act Repealed by Central Act 2 of 1999).

Cotton Ginning and Pressing Factories (Hyderabad Amendment) Act, 1954 (Hyderabad) (XXXIII of 1954) (Principal Act Repealed by Central Act 2 of 1999).

Maharashtra Cotton Seeds (Regulation of supply, distribution, sale and fixation of sale price) Act, 2009 (Mah. XIX of 2009).

The Pulgaon Cotton Mills Limited (Acquisition of Shares) Act, 1982 (Mah. XXXIV of 1982). COURTS—

Court of Wards Act, 1350 F (Hyderabad) (XII of 1350 F).

Bombay High Court (Letters Patent) Act, 1866 (Bom. XXIII of 1866).

Maharashtra Civil Courts Act, 1869 (XIV of 1869).

Maharashtra Revenue Jurisdiction Act, 1876 (X of 1876).

Court of Wards Act, 1899 (C. P. and Berar) (XXIV of 1899).

Mamlatdars' Courts Act, 1906 (Bom. II of 1906).

Provincial Small Cause Courts (Bombay Amendment) Act, 1930 (Bom. VI of 1930).

Provincial Small Cause Courts (Bombay Amendment) Act, 1932 (Bom. IX of 1932).

Presidency Small Cause Courts (Bombay Amendment) Act, 1933 (Bom. V of 1933).

Greater Bombay Laws and the Bombay High Court (Declaration of Limits) Act, 1945 (Bom. XVII of 1945).

Bombay Special Tribunal (Continuance) Act, 1946 (Bom. XXI of 1946).

Bombay City Civil Court Act, 1948 (Bombay XL of 1948).

Presidency Small Cause Courts (Bombay Amendment) Act, 1948 (Bom. XLIV of 1948).

Bombay High Court Letters Patents (Amendment) Act, 1948 (Bom. XLI of 1948).

Provincial Small Cause Courts (Bombay Amendment) Act, 1950 (Bom.LVII of 1950).

Presidency Small Cause Courts (Bombay Amendment) Act, 1952 (Bom. XVII of 1952).

Presidency Small Cause Courts (Bombay Amendment) Act, 1953 (Bom. LIX of 1953).

Provincial Small Cause Courts (Suits Validation) Act, 1955 (Bom. XVI of 1955).

Provincial Small Cause Courts (Bombay Amendment) Act, 1955 (Bom. XLIV of 1955).

Greater Bombay Laws and the Bombay High Court (Declaration of Limits) (Amendment) Act, 1956 (Bom. LVII of 1956).

Provincial Small Cause Courts (Bombay Unification and Amendment) Act, 1958 (Bom. LXXXVII of 1958).

Presidency Small Cause Courts (Bombay Amendment) Act, 1959 (Bom. XI of 1959).

Presidency Small Cause Courts (Maharashtra Amendment) Act, 1961 (Mah. XXXV of 1961).

Presidency Small Cause Courts (Maharashtra Amendment) Act, 1963 (Mah. XLI of 1963).

Presidency Small Cause Courts (Maharashtra Amendment) Act, 1966 (Mah. XV of 1966).

Presidency Small Cause Courts (Maharashtra Amendment) Act, 1976 (Mah. XIX of 1976).

Provincial Small Cause Courts and Presidency Small Cause Courts (Maharashtra Amendment) Act, 1984 (Mah. XXIV of 1984).

Civil Courts (Enhancement of Pecuniary Jurisdiction and Amendment) Act, 1977, Maharashtra (Mah. XLVI of 1977).

COURT FEES—

See "Civil Procedure and Practice".

CRIMINAL COURTS—

See "Criminal Law and Procedure".

CRIMINAL LAW AND PROCEDURE—

Maharashtra Prevention of Communal, Anti-Social and other Dangerous Activities Act, 1980 (Mah. VII of 1981).

Maharashtra Prevention of Dangerous Activities of Slum-lords, Bootleggers, Drugoffenders and Dangerous Persons Video Pirates, Sand Smugglers and persons engaged in Black-marketing of Essential Commodities Act, 1981 (Mah. LV of 1981).

Prevention of Thefts from Gold Mines Regulations, 1359 F (Hyderabad XXIII of 1359 F). Maharashtra Prevention of Gambling Act (IV of 1887).

Game Act, 1935 (C. P. and Berar) (XV of 1935).

Probation of Offenders Act, 1936 (C. P. and Berar) (I of 1936).

Probation of Offenders Act, 1953 (Hyderabad) (XII of 1953).

Coroners (Bombay Amendment) Act, 1954 (Bom. LXIX of 1954) (Principal Act Repealed by Mah. VIII of 2000).

Bombay Abolition of Whipping Act, 1957 (Bom. XXXIX of 1957).

Criminal Law Amendment (Bombay Amendment and Extension) Act, 1959 (Bom. XXIV of 1959).

Bombay Habitual Offenders Act, 1959 (Bom. LXI of 1959).

Maharashtra Government Servants Inquiries (Evidence of Corruption) Act, 1965 (Mah. XLIV of 1965).

Probation of Offenders (Maharashtra Amendment) Act, 1969 (Mah. XXXI of 1969).

Maharashtra Preventive Detention Act, 1970 (Mah. XVIII of 1970) (Remained in force from the 31-12-1969 upto and inclusive of 31-12-1974 and then expired).

Maharashtra Lokayukta and Upa-Lokayuktas Act, 1971 (Mah. XLVI of 1971).

Maharashtra Vexatious Litigation (Prevention) Act, 1971 (Mah. XLVIII of 1971).

Indian Criminal Law Amendment (Maharashtra Amendment) Act, 1975 (Mah. XXXVIII of 1975).

Maharashtra Taxation Laws Offences (Extension of Period of Limitation) Act, 1977 (Mah. XLIV of 1977).

Maharashtra Taxation Laws Offences (Extension of Period of Limitation) Act, 1982 (Mah. XXII of 1982).

Maharashtra Private Security Guards (Regulation of Employment and Welfare) Act, 1981 (Mah. LVIII of 1981).

Code of Criminal Procedure (Maharashtra Amendment) Act, 1976 (Mah. XXIII of 1976).

Code of Criminal Procedure (Maharashtra Amendment) Act, 1977 (Mah. I of 1978).

Code of Criminal Procedure (Maharashtra Amendment) Act, 1981 (Mah. XXXIV of 1981).

Code of Criminal Procedure (Maharashtra Second Amendment) Act, 1981 (Mah. LX of 1981).

Code of Criminal Procedure (Maharashtra Amendment) Act, 2007 (Mah. XXVII of 2007).

The Code of Criminal Procedure (Maharashtra Amendment) Act, 2018 (Mah. LXI of 2018).

The Code of Criminal Procedure (Maharashtra Amendment) Act, 2018 (Mah. LXXI of 2018).

Maharashtra Control of Organised Crime Act, 1999 (Mah. XXX of 1999).

Maharashtra Protection of Interest of Depositors (in Financial Establishments) Act, 1999 (Mah. XVI of 2000).

CRUELTY TO ANIMALS, PREVENTION OF—

See "Animals and Birds".

DEBT AND DEBTORS—

See "Agriculture".

DEVADASIS—

Maharashtra Devdasi System (Abolition) Act, 2005 (Mah. XXXIII of 2006).

DISEASES—

Pest—See "Agriculture".

Infectious Diseases—See "Public Health".

Epidemic—See "Public Health".

DISPLACED PERSONS—

Maharashtra Project Affected Persons Rehabilitation Act, 1999 (Mah. XI of 2001).

DISPOSITION OF PROPERTY—

See "Trusts and Trustees".

DISQU ALIFIC ATIONS, REMOVAL OF—

Maharashtra Legislature Members (Removal of Disqualifications) Act, 1956 (Bom. LII of 1956).

DISTRICT OFFICERS—

District Officers (Change of Designation and Construction of References) Act, 1950 (C. P. and Berar Act 35 of 1950).

DIVORCE—

See "Laws".

See "Personal Laws".

DOMESTIC—

Maharashtra Domestic Workers Welfare Board Act, 2008 (Mah. I of 2009).

DRUGS—

See "Medicine".

DUTY ON ELECTRICITY—

See "Revenue".

DUTY ON ENTERTAINMENTS—

See "Revenue".

EASEMENTS—

An Act to extend the Indian Easements Act, 1882 to Bombay (VIII of 1891).

EDUCATION—

Recognised Examinations Act, 1937 (C. P. and Berar) (X of 1937).

Bombay Secondary School Certificate Examination Act (Application to Scheduled Areas) Regulation, 1951 (Bom. Reg. II of 1950).

Hyderabad Public Libraries (Repeal) Act, 1961 (Mah. XL of 1961).

Maharashtra Secondary and Higher Secondary Education Boards Act, 1965 (Mah. XLI of 1965).

Maharashtra Public Libraries Act, 1967 (Mah. XXXIV of 1967).

Maharashtra Educational Institutions (Transfer of Management) Act, 1971 (Mah. XLIX of 1971).

Maharashtra College Teachers (Temporary Postponement of Confirmation) Act, 1973 (Mah. XXXIII of 1973).

Employees of Private Schools (Conditions of Service) Regulation Act, 1977 (Mah. III of 1978).

Maharashtra State Board of Technical Education Act, 1997 (Mah. XXXVIII of 1997).

Maharashtra State Council of Examinations Act, 1998 (Mah. I of 1999).

Maharashtra Prohibition of Ragging Act, 1999 (Mah. XXXIII of 1999).

Maharashtra Pre-School Centres (Regulation of Admission) (Repeal) Act, 2000 (Mah. XXXI of 2001).

Maharashtra Private Professional Educational Institutions (Reservation of seats for admission for Scheduled Castes, Scheduled Tribes, De-notified Tribes (*Vimukta Jatis*), Nomadic Tribes and Other Backward Classes) Act, 2006 (Mah. XXX of 2006).

Maharashtra Scheduled Castes, Scheduled Tribes, De-notified Tribes (Vimukta Jatis), Nomadic Tribes, Other Backward Classes and Special Backward Category (Regulation of Issuance and Verification of) Caste Certificate Act, 2000 (Mah. XXIII of 2001).

Maharashtra Educational Institutions (Regulation of Fees) Act, 2011 (Mah. VII of 2014).

Maharashtra Self-financed Schools (Establishment and Regulation) Act, 2012 (Mah. I of 2013).

Maharashtra Unauthorised Institutions and Unauthorised Courses of Study in Agriculture, Animal and Fishery Sciences, Health Sciences, Higher, Technical and Vocational Education (Prohibition) Act, 2013 (Mah. XX of 2013).

Maharashtra State Board of Nursing and Paramedical Education Act, 2013 (Mah. XXIII of 2013).

Maharashtra Unaided Private Professional Educational Institutions (Regilation of Admissions and fees) Act, 2015 (Mah. XXVII of 2015).

Maharashtra State Reservation (of seats for admission in educational institutions in the State and for appointments in the public services and posts under the State) for Socially and Educationally Backward Classes (SEBC) Act, 2018 (Mah. LXII of 2018).

ELECTION—

State Election Commissioner (Qualification and Appointment) Act, 1994 (Mah. XXVII of 1994).

ELECTION OFFENCES—

See "Criminal Procedure".

ELECTRICITY—

Indian Electricity (Bombay Amendment) Act, 1946 (Bom. XVIII of 1946).

Maharashtra Electricity (Special Powers) Act (XX of 1946).

Bombay Electricity (Special Powers) Act (Application to Scheduled Areas) Regulation, 1955 (Bom. Reg. I of 1955).

Indian Electricity (Maharashtra Amendment) Act, 1974 (Mah. LXIII of 1974).

Indian Electricity (Maharashtra Amendment) Act, 1976 (Mah. XLIV of 1976).

Indian Electricity (Maharashtra Amendment) Act, 1981 (Mah. XLV of 1981).

ENGINEER'S CERTIFICATE—

See "Certificates".

ENTERPRISES—

Maharashtra State Enterprises (Restructuring and Other Special Provisions) (Repeal) Act, 2007 (Mah. IX of 2007).

ENTERTAINMENT DUTY—

See "Revenue".

EPIDEMIC DISEASES—

See "Public Health".

ERADICATION—

Maharashtra Prevention and Eradication of Human Sacrifice and Other Inhuman, Evil and *Aghori* Practices and Black Magic Act, 2013 (Mah. XXX of 2013).

ESSENTIAL COMMODITIES—

See "Trade".

ESSENTIAL SERVICES—

Maharashtra Essential Services Maintenance Act, 2017 (Mah. XVIII of 2018).

ESSENTIAL SUPPLIES—

See "Trade".

ESTABLISHMENTS—

See "Trade".

EVACUEES—

Bombay Evacuee Interest (Separation) Validating and Supplementary Act, 1953 (Bom. XXIX of 1953).

Evacuee Interest (Separation) Supplementaly Act, 1953 (Hyderabad) (VII of 1953).

Maharashtra Evacuee Interest (Separation) Supplementary Act, 1960 (Mah. XXV of 1960).

EVICTION FROM GOVERNMENT PREMISES—

See "Government Property".

FACTORIES—

See "Cotton".

FAIRS AND MARKETS—

See "Trade".

FERRIES—

See "Transport".

FINANCE—

Central Provinces and Berar Finance (Bombay Repeal) Act, 1958 (Bom. XLVII of 1958).

Maharashtra Finance Commission (Miscellaneous Provisions) Act, 1994 (Mah. XXIII of 1994).

FINE ART LITHO WORKS—

Shivraj Fine Art Litho Works (Acquisition and Transfer of Undertaking) Act, 1984 (Mah. XXXVI of 1984).

FISHERIES—

Maharashtra Fisheries Act, 1960 (Mah. I of 1961).

Maharashtra Marine Fishing Regulation Act, 1981 (Mah. LIV of 1981).

FORESTS—

Indian Forest (Bombay Amendment) Act, 1948 (Bom. LXII of 1948).

Indian Forest (Bombay Amendment) Act, 1953 (Bom. XXV of 1953).

Indian Forest (Bombay Amendment) Act, 1955 (Bom. XXIV of 1955).

Indian Forest (Bombay Amendment) Act, 1956 (Bom. X of 1956).

Indian Forest (Bombay Second Amendment) Act, 1956 (Bom. XVII of 1956).

Indian Forest (Bombay Amendment) Act, 1957 (Bom. XXVI of 1957).

Indian Forest (Maharashtra Unification and Amendment) Act, 1960 (Mah. VI of 1961).

Indian Forest (Maharashtra Amendment) Act, 2013 (Mah. XXI of 2015).

Forest (Maharashtra Amendment) Act, 1964 (Mah. XV of 1965).

Forest (Maharashtra Amendment) Act, 1968 (Mah. XXVII of 1968).

Forest (Maharashtra Amendment) Act, 1984 (Mah. VII of 1985).

Forest (Maharashtra Second Amendment) Act, 1984 (Mah. XXIII of 1984).

Maharashtra Forest Produce (Regulation of Trade) Act, 1969 (Mah. LVII of 1969).

Maharashtra Forest Produce (Regulation of Trade) (Amendment) Act, 2014 (Mah. XXXII of 2014).

Maharashtra Private Forest (Acquisition) Act, 1975 (Mah. XXIX of 1975).

Maharashtra Supply of Forest Produce by Government (Revision of Agreements) Act, 1982 (Mah. XVI of 1983).

FOREIGN IMMIGRANTS—

Foreign Immigrants Regulation (C. P. and Berar Act XI of 1812).

FORWARD CONTRACTS—

See "Contracts".

FRANCHISE LOCAL AUTHORITIES ADULT—

See "Local Authorities".

FUND—

Improvement Fund—See "Agriculture".

Advocate Welfare Fund—See "Advocate".

Contingency Fund—See "Contingency Fund".

Gandhi National Memorial Fund—See "Local Authorities".

Labour Welfare—See "Labour".

Post War Reconstruction—See "Local Authorities".

Local Fund Audit—See "Local Authorities".

FUND, CONTINGENCY—

Maharashtra Contingency Fund Act (Bom. XLVI of 1956).

FUND, LOCAL DEVELOPMENT—

See "Local Authorities".

FUND, STATUTORY—

Maharashtra Statutory Funds Act (LXXI of 1959).

GAMBLING—

See "Criminal Law and Procedure".

GAS-

Bombay Gas Supply Act, 1939 (Bom. IX of 1939).

Gas Companies Act, 1863 (Bom. V of 1863).

GENERAL CLAUSES ACT—

See "Laws".

GOVERNMENT PROPERTY—

Railways Government Irrigation Sources and Other Public Buildings Protection Act, 1321 F (Hyderabad) (I of 1321 F).

Maharashtra Government Premises (Eviction) Act (II of 1956).

Maharashtra Vacant Lands (Prohibition of Unauthorised Occupation and Summary Eviction) Act, 1975 (Mah. LXVI of 1975).

Maharashtra Vacant Lands (Further Interim Protection to Occupiers from Eviction and Recovery of Arrears of Rent) Act, 1980 (Mah. XVI of 1980).

Maharashtra Gunthewari Developments (Regularisation, Upgradation and Control)

Act, 2001 (Mah. XXVII of 2001).

GROUND WATER—

Maharashtra Ground Water (Development and Management) Act, 2009 (Mah. XXVI of 2013).

GUARANTEES—

Bombay State Guarantees (Repeal) Act, 1965 (Mah. XXI of 1965).

HABITUAL OFFENDERS—

See " Criminal Law and Procedure".

HEALTH VISITORS—

See " Medical Profession".

HEREDITARY OFFICES—

See " Land Tenures".

HIGH COURT—

See " Courts".

HIGHWAYS—

See " Communications".

HINDU CEREMONIAL EMOLUMENTS—

See "Personal Law".

HINDU HEIRS—

See " Personal Law".

HINDU WOMEN'S RIGHTS TO PROPERTY—

See "Personal Law".

HOLIDAYS—

Weekly Holiday (Substitution of Sunday For Friday) Act, 1950 (Hyderabad) (VIII of 1950).

HOME GUARDS—

Maharashtra Home Guards Act (III of 1947).

HOMOEOPATHY—

See " Medical Profession".

HORTICULTURE—

Maharashtra Horticulture Development Corporation Act, 1984 (Mah. XXV of 1984).

HOTELS AND LODGING HOUSES—

See "Rent Control".

HOUSING BOARD—

Maharashtra Housing and Area Development Act, 1977 (Mah. XXVIII of 1977).

HUSBAND AND WIFE—

Registrar General of Births, Deaths and Marriages Act (Hyderabad) (XVIII of 1953).

Maharashtra Regulation of Marriage Bureaus and Registration of Marriages Act, 1998 (Mah. XX of 1999).

IDENTIFICATION OF PRISONERS—

See "Prisons, Prisoners and Reformatories".

IMPROVEMENT OF LAND SCHEMES—

See " Agriculture".

IMPROVEMENT OF LIVESTOCK—

See " Animals and Birds".

INAMS—

See " Land Tenure".

INCOME-TAX (BOMBAY REPEAL), HYDERABAD AGRICULTURAL—

See "Revenue".

See " Agriculture".

INDUSTRIAL DISPUTES—

See "Labour".

INDUSTRIAL RELATIONS—

See "Labour".

INDUSTRIAL WORKERS DEBT—

See "Labour".

INDUSTRY—

Maharashtra Relief Undertakings (Special Provisions) Act (XCVI of 1958).

Maharashtra Khadi and Village Industries Act (XIX of 1960).

Relief Undertakings—See "Labour"

Maharashtra State Aid to Industries Act, 1960 (Mah. XVII of 1960).

Maharashtra Industrial Development Act, 1961 (Mah. Ill of 1962).

INSOLVENCY—

Insolvency Act, 1351 F (Hyderabad) (VIII of 1351 F).

Presidency Towns Insolvency (Bombay Amendment) Act, 1933 (Bom. XX of 1933).

Presidency Towns Insolvency and the Provincial Insolvency (Bombay Amendment) Act, 1939(Bom. XVof 1939).

Presidency Towns Insolvency (Bombay Amendment) Act, 1948 (Bom. LI of 1948).

INTEREST—

Interest (Extension) Act, 1960 (Mah. XXIII of 1960).

IRRIGATION—

Maharashtra Irrigation Act, 1976 (Mah. XXXVIII of 1976).

Maharashtra Management of Irrigation Systems by Farmers Act, 2005 (Mah. XXIII of 2005).

JAGIRS—

See " Land Tenure".

JAILS, CIVIL—

See "Prisons, Prisoners and Reformatories".

JUDICIAL OFFICERS, PROTECTION—

See "Public Service".

JUDICIAL PROCEEDINGS (REGULATION OF REPORTS)—

See "Civil Procedure and Practice".

KAULI KATUBAN TENURE—

See " Land Tenures".

KAZI—

Kazis (Maharashtra Amendment) Act, 1978 (Mah. XXI of 1978).

KHAR LANDS—

Maharashtra Khar Land Development Act, 1979 (Mah. XI of 1979).

KHOTI TENURE-ABOLITION OF —

See " Land Tenure".

LABOUR—

Adjustment and Liquidation of Industrial Workers' Debt Act, 1936 (C. P. and Berar) (V of 1936).

Maharashtra Industrial Relations Act (XI of 1947).

Labour Housing Act, 1952 (Hyderabad) (XXXVI of 1952).

Maharashtra Labour Welfare Fund Act (XL of 1953).

Industrial Disputes (Appellate Tribunal) (Bombay Amendment) Act, 1953 (Bom. XVI of 1953).

Industrial Disputes (Appellate Tribunal) (Bombay Amendment) Act, 1954 (Bom. LXV of 1954).

Industrial Employment (Standing Orders) (Bombay Amendment) Act, 1956 (Bom. XXXVI of 1956).

Indian Trade Unions (Bombay Amendment) Act, 1956 (Bom. XXXII of 1956).

Central Provinces Industrial Disputes Settlement (Commencement of Certain Amending Provisions and Validation) Act, 1958 (Bom. LIX of 1958).

Industrial Employment (Standing Orders) (Bombay Amendment) Act, 1958 (Bom. XXI of 1958).

Payment of Bonus (Maharashtra Amendment) Act, 2010 (Mah. XVI of 2010).

Payment of Wages (Bombay Amendment) Act, 1953 (Bom. LXII of 1953).

Payment of Wages (Bombay Amendment) Act, 1954 (Bom. LXX of 1954).

Payment of Wages (Bombay Amendment) Act, 1955 (Bom. XLVIII of 1955).

Payment of Wages (Bombay Amendment) Act, 1956 (Bom. XXXVII of 1956).

Payment of Wages (Bombay Amendment and Validation) Act, 1959 (Bom. VIII of 1960). Payment of Wages (Maharashtra Provision for Uniform Application and Amendment)

Act, 1960 (Mah. XIII of 1961).

Payment of Wages (Maharashtra Amendment) Act, 1961 (Mah. XLII of 1961).

Payment of Wages and Minimum Wages (Maharashtra Amendment) Act, 2010 (Mah. XVI of 2010).

Minimum Wages (Maharashtra Amendment) Act, 1960 (Mah. X of 1961).

Minimum Wages (Maharashtra Amendment) Act, 1963 (Mah. Ill of 1963).

Trade Unions and Bombay Industrial Relations (Amendment) Act, 1967 (Mah. Ill of 1968). Maharashtra Mathadi, *Hamal* and Other Manual Workers (Regulation of Employment and Welfare) Act, 1969 (Mah. XXX of 1969).

Maharashtra Recognition of Trade Unions and Prevention of Unfair Labour Practices Act, 1971 (Mah. I of 1972).

Maharashtra Industrial Relations (Validation of Certain Proceedings) Act, 1973 (Mah. X of 1973).

Industrial Employment (Standing Orders) (Maharashtra Amendment) Act, 1974 (Mah. LIV of 1974).

Industrial Disputes (Maharashtra Amendment) Act, 1974 (Mah. LVI of 1974).

Minimum Wages (Maharashtra Amendment) Act, 1975 (XXV of 1976).

Industrial Disputes (Maharashtra Amendment) Act, 1981 (Mah. Ill of 1982).

Recovery and Penalty Provisions of Maharashtra Ordinance No. Ill of 1975 Continuance Act, 1975 (Mah. XXXIX of 1975).

Industrial Disputes and the Maharashtra Unemployment Allowance Payment to Workmen in Factories (for Temporary Period) (Amendment) Act, 1981 (Mah. XXII of 1981).

LAND ACQUISITION—

See " Acquisition and Requisition".

LAND TENURES—

Abolition of Jagirs Regulation, 1358F (Hyderabad) (LXIX of 1358F).

Jagirs (Commutation) Regulation, 1359F (Hyderabad) (XXV of 1359F).

Bombay Rent-free Estates Act, 1852 (XI of 1852).

Exemptions from Land Revenue (No. 1) Act, 1863 (Bom. II of 1863).

Exemptions from Land Revenue (No. 2) Act, 1863 (Bom. VII of 1863).

Maharashtra Hereditary Offices Act (III of 1874).

Mahul Creek (Extinguishment of Rights) Act, 1922 (Bom. IV of 1922).

West Khandesh (Mehwassi Estates) Regulation, 1949 (Bom. Reg. I of 1949).

Maharashtra Bhagdari and Narwadari Tenure Abolition Act (XXXII of 1949).

Maharashtra Khoti Abolition Act (VI of 1950).

District Officers (Change of Designation of Construction of References) Act, 1950 (C. P. and Berar Act XXXV of 1950).

Maharashtra Paragana and Kulkami Watans (Abolition) Act (LX of 1950).

Abolition of Proprietory Rights (Estates, Mahals, Alienated Lands) Act, 1951 (M. P.) (I of 1951).

Jagirdars Debts Settlement Act, 1952 (Hyderabad) (XII of 1952).

Abolition of Cash Grants Act, 1952 (Hyderabad) (XXXIII of 1952).

Bombay Land Tenures Abolition (Amendment) Act, 1953 (Bom. XXXVIII of 1953).

Maharashtra Personal Inams Abolition Act (XLII of 1953).

Bombay Merged Territories (Ankadia Tenure Abolition) Act (XLIII of 1953).

Maharashtra Kauli and Katuban Tenures (Abolition) Act (XLIV of 1953).

Maharashtra Land Tenures Abolition (Recovery of Records) Act (L of 1953).

Maharashtra Service Inams (Useful to Community) Act (LXX of 1953).

Maharashtra Merged Territories (Janjira and Bhor) Khoti Tenure Abolition Act, 1953 LXXI of 1953).

Bombay (Okhamandal Salami Tenure Abolition) Act, 1953 (Bom. I of 1954).

Maharashtra Merged Territories and Areas (Jagirs Abolition) Act (XXXIX of 1954).

Maharashtra Bhil Naik Inams Abolition Act (XXI of 1955).

Abolition of Inams Act, 1955 (Hyderabad) (VIII of 1955).

Maharashtra Merged Territories Miscellaneous Alienation Abolition Act (XXII of 1955).

Maharashtra Shilotri Rights (Kolaba) Abolition Act (XLVII of 1955).

Bombay Land Tenure Abolition (Amendment) Act, 1956 (Bom. XL of 1956).

Maharashtra Shetgi Watan (Ratnagiri) Abolition Act (II of 1957).

Bombay Land Tenure Abolition Law (Amendment) Act, 1958 (Bom. LVII of 1958).

Maharashtra Inferior Village Watans Abolition Act (I of 1959).

Maharashtra Bandhijama Udhad and Ugadia Tenures Abolition Act (XXXV of 1959).

Hyderabad Abolition of Inams (Amendment) Act, 1959 (Bom. LXIV of 1959).

West Khandesh Mehwassi Estates (Proprietory Rights Abolition) Regulation, 1961 (Mah. Reg. I of 1962).

Maharashtra Miscellaneous Alienations (in Hyderabad Enclaves) Abolition Act, 1965 (Mah. LVI of 1965).

Bombay City (Inami and Special Tenures) Abolition and Maharashtra Land Revenue Code (Amendment) Act, 1969 (Mah. XLIV of 1969).

LAWS—

General Clauses Act, 1308 F (Hyderabad) (III of 1308 F).

Laws Act, 1875 (C. P.) (XX of 1875).

Maharashtra General Clauses Act (I of 1904).

Laws Provincial Act, 1941 (Berar) (XV of 1941).

Bombay Repealing and Amending Act, 1949 (Bom. LIII of 1949).

General Clauses (Amendment and Supplementary) Act, 1950 (XVIII of 1950).

Bombay (Second) Repealing and Amending Act, 1951 (Bom. XXXIX of 1951).

Akrani Mahal (Application of Laws) Regulation, 1952 (Bom. Reg. I of 1952).

Application of Central Acts, 1952 (Hyderabad) (XVIII of 1952).

Bombay Repealing and Amending Act, 1954 (Bom. XXI of 1954).

Bombay Repealing and Amending Act, 1955 (Bom. II of 1955).

Bombay (Second) Repealing and Amending Act, 1955 (Bom. XXIX of 1955).

Bombay Repealing and Amending Act, 1959 (Bom. LXIII of 1959).

Maharashtra Construction of References to "District Magistrate" in the Hyderabad Area (Repeal) Act, 1960 (Mah. XIV of 1960).

Maharashtra Deletion of the Term "famine" (from Laws applicable to the State) Act, 1963 (Mah. XL of 1963).

Maharashtra Local Authorities (Postponement of Elections for the Duration of the Emergency) (Repeal) Act, 1963 (Mah. XXII of 1963).

Maharashtra Repealing and Amending Act, 1963 (Mah. XXVI of 1963).

Surgeon General with Government, etc. (Change in Designation) Act, 1972 (Mah. Ill of 1973).

Maharashtra Director of Prohibition and Excise (Change of Designation) Act, 1973 (Mah. LII of 1973).

Maharashtra Repealing and Amending Act, 1976 (Mah. XI of 1976).

Maharashtra Repealing and Amending Act, 1991 (Mah. XXII of 1991).

LAWS, EXTENSION OF—

Dakkhan and Khandesh (Puna, Ahmednagar and Khandesh Districts), Bombay Regulation XXIX of 1827.

Satara, Solapur and Sourthen Maratha Country Laws Act, 1863 (Bom. Ill of 1863).

Edulabad and Warangaon Parganas Laws Act, 1866 (Bom. XIV of 1866).

Peint Laws Act, 1894 (Bom. II of 1894).

Maharashtra Merged States (Laws) Act (IV of 1950).

Bombay Extension of Laws to Non-Scheduled (Partially Excluded) Areas Act, 1954 (Bom. LXVIII of 1954).

LAWS FOR CITY OF BOMBAY ONLY—

City of Bombay (Building Restriction) Act, 1949 (Bom. XLIV of 1949).

Bombay Queen Victoria Statue Site (and adjoining land utilisation for construction of Satellite

Telecommunications Exchanges of the Overseas Communications Service) Act, 1968.

(Mah. V of 1968).

LEGAL PRACTITIONERS—

Legal Practitioners (Bombay Amendment) Act, 1948 (Bom. XLV of 1948).

LEGISLATURE — STATE—

Bombay Prohibition of Simultaneous Membership Act, 1957 (Bom. XII of 1957).

Maharashtra State Legislature Proceedings (Protection of Publication) Act, 1969 (Mah. XXXII of 1969).

LEGISLATURE MEMBERS, SALARIES AND ALLOWANCES AND PENSION—

See "Salaries and Allowances and Pension".

LEGISLATURE MEMBERS—REMOVAL OF DISQUALIFICATION—

See "Disqualifications, Removal of".

LETTERS PATENTS

See "Courts".

LIFTS—

Maharashtra Lifts, Escalators and Moving Walks Act, 2017 (Mah. XV of 2018).

LIMITATION—

See "Civil Procedure and Practice".

LITHO WORKS-

Shivraj Fine Art Litho Works (Acquisition and Transfer of Undertaking) Act, 1984 (Mah. XXXVI of 1984).

LOANS-

AGRICULTURISTS—

See " Agriculture".

LOCAL AUTHORITIES—

Mumbai Municipal Corporation Act (III of 1888).

City of Bombay Municipal (Supplementary) Act, 1888 (Bom. XII of 1888).

Bombay Municipal Servants Act, 1890 (Bom. V of 1890).

Maharashtra Local Fund Audit Act (XXV of 1930).

Improvement Trust Act, 1936 (Nagpur) (XXXVI of 1936).

Collection of Statistics Act, 1939 (C. P. and Berar) (XXXIV of 1939).

Local Authorities Loans (Bombay Amendment) Act, 1942 (Bom. II of 1942).

Maharashtra Municipal Corporations Act (LIX of 1949).

Н 1077—7а

Bombay Local Authorities Census Expenses Contribution Act, 1950 (Bom. XXIII of 1950).

Local Bodies Services (Commission) Act, 1950 (M. P.) (XXXVII of 1950).

Local Authorities (Electoral Offences) Act, 1952 (M. P.) (XI of 1952).

Gandhi National Memorial Fund (Local Authorities Donations) Act, 1953 (Bom. XVII of 1953).

Local Government Services (Declaration as State Civil Service) Act, 1956 (Hyderabad) (XX of 1956).

Maharashtra Village Panchayats Act (III of 1959).

Maharashtra Zilla Parishads and Panchayat Samitis Act, 1961 (Mah. V of 1962).

Maharashtra Municipal Councils, *Nagar Panchayats* and Industrial Townships Act, 1965 (Mah. XL of 1965).

Maharashtra Extension of Municipal Limits (Validation) Act, 1967 (Mah. V of 1967).

Maharashtra Municipal Councils (Postponement of Elections due to Scarcity Conditions in the State) Act, 1972 (Mah. XLVI of 1972).

Mumbai Metropolitan Region Development Authority Act, 1974 (Mah. IV of 1975).

Mumbai Metropolitan Region Specified Commodities Markets (Regulation of Location) Act, 1983 (Mah. XLII of 1983).

Maharashtra Village Panchayats (Postponement of Elections during the Emergency) Act, 1975 (Mah. XLVIII of 1975).

Maharashtra Village Panchayats (Temporary further Postponement of Elections due to Elections to Maharashtra Legislative Assembly) Act, 1980 (Mah. XIV of 1980).

Maharashtra Village Panchayats (Temporary Postponement of Elections due to Elections Preparation of Revised Assembly Roll) Act, 1983 (Mah. XLVI of 1983).

Maharashtra Municipal Councils and Municipal Corporations (Postponement of Elections during the Emergency) Act, 1975 (Mah. LV of 1975).

Maharashtra Village Panchayats (Postponement of Elections due to ensuing General Elections to the State Legislative Assembly, Zilla Parishads and Panchayat Samitis) Act, 1977 (Mah. LI of 1977).

Municipal Councils and Municipal Corporations (Postponement of Elections due to ensuing General Elections to the State Legislative Assembly) Act, 1977 (Mah. XLVIII of 1977).

Municipalities (Postponement of Elections due to ensuing General Elections to Municipal Councils) Act, 1978 (Mah. XXVI of 1978).

Municipal Corporation (Bombay) (Revision of Election Rules and Further Extension of Term) Act, 1978 (Mah. XVII of 1978).

Maharashtra Village Panchayats (Temporary Postponement of Elections due to Election of Lok Sabha) Act, 1980 (Mah. Ill of 1980).

Maharashtra Zilla Parishads and Panchayat Samitis Temporary Postponements of Elections (of the President, Vice-President and Chairman of Subjects Committies of certain Zilla Parishads and Chairman and Deputy Chairman of certain Panchayat Samitis due to ensuing General Elections to the State Legislative Assembly and certain other elections) Act, 2004 (Mah. VII of 2004).

Maharashtra Municipal Corporations Temporary Postponement of Elections (of the Mayors and Deputy Mayors due to ensuing general elections to the State Legislative Assembly and certain other elections) Act, 2004 (Mah. XVII of 2004).

Maharashtra Water Conservation Corporation Act, 2000 (Mah. Ill of 2001).

Maharashtra Water Resources Regulatory Authority Act, 2005 (Mah. XVIII of 2005).

LOCATION—

Truck Terminal (Regulation of Location) Act, 1997 (Mah. XXXI of 1997).

LOTTERIES—

See "Revenue".

LUNACY—

Indian Lunacy, Bombay District Municipal and Bombay Municipal Boroughs (Amendment) Act, 1936 (Bom. XV of 1936).

MAMLATDARS COURTS—

See " Courts ".

MARKETS, AGRICULTURAL PRODUCE—

See "Agriculture".

MARKETS AND FAIRS—

See "Trade".

Mumbai Metropolitan Region Specified Commodities Markets (Regulation of Location) Act, 1983 (Mah. XLII of 1983).

MARRIAGES—

See "Husband and Wife" and "Personal law".

MEDICAL PROFESSION—

Maharashtra Homoeopathic and Biochemic Practitioners' Act (XII of 1960).

Maharashtra Medical Practitioners' Act, 1961 (Mah. XXVIII of 1961).

Indian Medical Degrees (Maharashtra Extension and Provision for Uniformity) Act, 1961 (Mah. XLIVof 1961).

Maharashtra Medical Council Act, 1965 (Mah. XLVI of 1965).

Maharashtra Nurses Act, 1966 (Mah. XL of 1966).

Maharashtra Medical and Dental Colleges Admissions (Regulation and Abolition of All India Quota) Act, 2003 (Mah. XV of 2003).

Maharashtra State Council for Occupational Therapy and Physiotherapy Act, 2002 (Mah. II of 2004).

MEDICINE—

Maharashtra Opium Smoking Act (XX of 1936).

Drugs (Control) Act, 1949 (C. P. and Berar) (XLVII of 1949).

Maharashtra Drugs (Control) Act (XI of 1960).

Maharashtra Medicare Service Persons and Medicare Service Institutions (Prevention of Violence and Damage or Loss to Property) Act, 2010 (Mah. XI of 2010).

MEMBERSHIP—PROHIBITION OF SIMULTANEOUS—

See "Legislature—State".

MERGED AREAS—

Bombay Merged Areas Enclaves and Specified Areas (Amendment of Laws) Act, 1950 (Bom. XXII of 1950).

MERGED TERRITORIES—

See "Land Tenure".

MILITARY AUTHORITY—

Military Authority (Assistance to Marching Troops) (Bombay Regulation XXII of 1827).

MINES AND MINERALS—

Maharashtra Abolition of Subsisting Proprietary Rights to Mines and Minerals in Certain Lands Act, 1985 (Mah. XVI of 1985).

MINISTERS SALARIES, ALLOWANCES AND PENSION—

See "Salaries, Allowances and Pension".

MONEY LENDING—

Bombay Non-Agriculturists' Loans Act, 1928 (Bom. III of 1928).

Maharashtra Money-Lending (Regulation) Act, 2014 (Mah. VIII of 2014).

MONUMENTS AND MEMORIALS—

Maharashtra Ancient Monuments and Archaeological Sites and Remains Act, 1960 (Mah. XII of 1961).

MOTOR VEHICLES—

See "Transport".

MOTOR VEHICLES TAX—

See "Revenue".

MOTOR SPIRIT—

See "Revenue".

MUNICIPAL CORPORATIONS—

See " Local Authorities".

MUNICIPAL DEBENTURES—

See " Local Authorities ".

MUNICIPAL PROPERTY TAX—

See "Revenue".

MUNICIPAL SERVANTS—

See "Local Authorities".

MUNICIPALITIES—

See " Local Authorities ".

MUSEUMS—

Chhatrapati Shivaji Maharaj Vastu Sangrahalaya (Museum) Act, 1909 (Bom. III of 1909).

Mahatma Phule Vastu Sangrahalaya, Pune Act, 1968 (Mah. XVIII of 1968).

NON-AGRICULTURISTS—

See "Money-Lending".

Shore Nuisances (Bombay and Kolaba) Act, 1853 (XI of 1853).

NURSES—

See " Medical Profession ".

NURSERIES—

Maharashtra Fruit Nurseries (Regulation) Act, 1969 (Mah. XLV of 1969).

NURSING HOMES—

See " Public Health ".

OFFICIAL LANGUAGE—

Languages of Judgement and Orders Act, 1950 (Hyderabad) (XXIX of 1950).

Maharashtra Official Languages Act, 1964 (Mah. V of 1965).

OPIUM—

See " Medicine ".

PANDHARPUR—

Pandharpur Development Authority Act, 2009 (Mah. XV of 2009).

PARTNERSHIP—

Partnership (Maharashtra Amendment) Act, 1973 (Mah. XXV of 1973).

Partnership (Maharashtra Amendment) Act, 1984 (Mah. XXIX of 1984).

PAYMENT OF BONUS—

See "LABOUR".

PAYMENT OF WAGES—

See "LABOUR".

PERSONAL LAW—

Hindu Widows' Marriage Act, 1347 F (Hyderabad) (I of 1347 F).

Bombay Hindu Heirs Relief Act, 1866 (Bom. VII of 1866).

Invalidation of Hindu (Ceremonial Emoluments) Act, 1926 (Bom. XI of 1926).

Hindu Women's Rights to Property (Extension to Agricultural Lands) Act, 1942 (C. P. and Berar) (VI of 1942).

Maharashtra Hindu Places of Public Worship (Entry Authorisation) Act (XXXI of 1956).

Hindu Succession (Maharashtra Amendment) Act, 1994 (Mah. XL of 1994).

PEST AND DISEASES—

See "Agriculture".

PILGRIMS, PROTECTION OF—

Protection of Pilgrims Act, 1887 (Bom. II of 1887).

Eradication of Unfair Activities at Pandharpur Temples (for providing better facilities for worship) Act, 1980 (Mah. XI of 1980).

PLEADERS—

Bombay Pleaders Act, 1920 (Bom. XVII of 1920).

POLICE—

Police (Incitement to Disaffection) (Maharashtra Amendment) Act, 1983 (Mah. XXIII of 1983).

Maharashtra Police Act (XXII of 1951).

Maharashtra State Reserve Police Force Act (XXXVIII of 1951).

Police (Incitement of Disaffection) (Bombay Extension and Amendment) Act, 1958 (Bom. LXXVII of 1958).

Maharashtra State Commissioners of Police Act (LVI of 1959).

Maharashtra Police Officers (Change in Designation) Act, 1962 (Mah. XLVI of 1962).

Bombay Village Police (Appointments and Disciplinary Action by Sub-Divisional Magistrates Validation) Act, 1967 (Mah. VI of 1967).

POST OFFICE—

Indian Post Office (Bombay Amendment) Regulation, 1942 (Bom. Reg. I of 1942).

PRESIDENCY SMALL CAUSE COURTS—

See "Courts".

PRESIDENCY TOWN INSOLVENCY—

See "Insolvency".

PRESS, PRINTING AND PUBLICATIONS—

Press and Registration of Books (Bombay Amendment) Act, 1948 (Bom. LXI of 1948).

Press and Registration of Books (Bombay Amendment) Act, 1951 (Bom. VI of 1951).

Press and Registration of Books (Extension of Bombay Amendment) Act, 1960 (Mah. VII of 1961).

PREVENTION OF ADULTERATION—

See "Public Health".

PREVENTION OF-

BEGGING—See "Beggar".

Defecement of Property Act, 1995 (Mah. VIII of 1995).

Vetti and Beggar-See "Beggar".

Cruelty to animals-See "Animals and Birds".

Fragmentation and Consolidation of Holdings-See "Agriculture".

Gambling-See " Criminal Law and Procedure ".

Thefts from Gold mines-See "Criminal Law and Procedure".

Adulteration (C. P. & Berar) See "Criminal Law and Procedure".

Food Adulteration-See "Criminal Law and Procedure".

Water Pollution-See "Pollution".

PRIMARY EDUCATION—

See " Education ".

PRISONS, PRISONERS AND REFORMATORIES—

Civil Jails Act, 1874 (Bom. II of 1874).

State Prisoners (Regulation) Act, 1818 (Bengal) (III of 1818).

Identification of Prisoners (Bombay Amendment) Act, 1922 (Bom. XI of 1922).

Identification of Prisoners (Bombay Second Amendment) Act, 1935 (Bom. XXI of 1935).

Identification of Prisoners (Bombay Amendment) Act, 1948 (Bom. XV of 1948).

Prisoners (Bombay Amendment) Act, 1949 (Bom. VII of 1949).

Prisons (Bombay Amendment) Act, 1953 (Bom. XXVII of 1953).

Identification of Prisoners (Bombay Amendment) Act, 1953 (Bom. LVIII of 1953).

Prisons (Bombay Amendment) Act, 1959 (Bom. XLV of 1959).

Prisoners (Bombay Extension and Amendment) Act, 1958 (Bom. XV of 1959).

Prisons and Civil Jails (Bombay Extension, Unification and Amendment) Act, 1958 (Bom. XXIII of 1959).

Probation of Offenders (Maharashtra Amendment) Act, 1969 (Mah. XXXI of 1969).

Identification of Prisoners (Maharashtra Provision for Uniform Application and Amendment) Act, 1970 (Mah. XXXV of 1970).

PRIVATE PROPERTY-ADMINISTRATION OF—

Sir Dinshaw Maneckjee Petit Act, 1893 (VI of 1893).

Sir Jamsetjee Jeejeebhoy Baronetcy Act, 1925 (X of 1925).

Maharashtra Baronetcies (Amendment) Act, 1974 (Mah. XXVIII of 1974).

PROBATION OF OFFENDERS—

See " Criminal Law and Procedure ".

PROHIBITION—

Power Alcohol Act, 1350 F (Hyderabad) (XI of 1350 F).

Maharashtra Prohibition Act (XXV of 1949).

Maharashtra Molasses (Control) Act (XXXVIII of 1956).

PROTECTION OF DEBTORS—

Protection of Debtors Act (C. P. and Berar) (C. P. and Berar IV of 1937).

PROTECTION OF PILGRIMS—

See "Pilgrims".

PROVINCIAL SMALL CAUSE COURTS—

See " Courts ".

PROVINCIAL MUNICIPAL CORPORATION—

See "Local Authorities".

PROVINCIAL COLLECTION OF TAXES—

See "Revenue".

PUBLIC AUTHORITIES-SEALS OF—

See "Seals".

PUBLIC CONVEYANCES—

See "Transport".

PUBLIC HEALTH—

Tuberculosis Sanatorium (Regulation of Buildings) Act, 1948 (C. P. and Berar) (XIII of 1948).

Maharashtra Nursing Homes Registration Act (XV of 1949).

Infectious Diseases Act, 1950 (Hyderabad) (XII of 1950).

Epidemic Diseases (Bombay Amendment) Act, 1953 (Bom. XII of 1953).

Lepers (Maharashtra Repeal) Act, 1983 (Mah. XL of 1983).

Maharashtra Slum Areas (Improvement, Clearance and Re-development) Act, 1971 (Mah. XXVIII of 1971).

Maharashtra Non-biogradable Garbage (Control) Act, 2006 (Mah. X of 2006).

PUBLIC SERVICES—

Judicial Officers Protection (Extension to Hyderabad and Saurashtra Areas of Bombay State) Act, 1958 (Bom. LXXVIII of 1958).

Maharashtra Public Services (Subordinate) Selection Boards (Repeal) Act, 1983 (Mah. XXXIV of 1983).

Maharashtra State Public Services (Reservation for Scheduled Castes, Scheduled Tribes, De-notified Tribes (*Vimukta Jatis*), Nomadic Tribes, Special Backward Category and Other Backward Classes) Act, 2001 (Mah. VIII of 2004).

Maharashtra Right to Information (Repeal) Act, 2005 (Mah. I of 2006).

Maharashtra Government Servants Regulation of Transfers and Prevention of Delay in Discharge of Official Duties Act, 2005 (Mah. XXI of 2006).

PUBLIC TRUSTS—

See "Trusts and Trustees".

RACE COURSES—

Maharashtra Race Courses Licensing Act (III of 1912).

RATIONING—

Bombay Rationing (Preparatory and Continuance) Measures Act, 1947 (Bom. LVIII of 1947).

RECORDS—

Destruction of Useless Records Act, 1305 F (Hyderabad) Act 3 of 1305 F.

Maharashtra Public Records Act, 2005 (Mah. IV of 2006).

REFUGEES—

C. P. and Berar Refugees Rehabilitation (Loans) (Mah. Repeal) Act, 1962 (Mah. XI of 1963).

REGISTRATION OF BOOKS—

See "Press, Printing and Publications."

REGISTRATION OF DOCUMENTS—

Indian Registration (Bombay Amendment) Act, 1929 (Bom. V of 1929).

Indian Registration (Bombay Amendment) Act, 1930 (Bom. XVII of 1930).

Indian Registration (Bombay Amendment) Act, 1933 (Bom. XVIII of 1933).

Indian Registration (Bombay Amendment) Act, 1938 (Bom. XXIV of 1938).

Transfer of Property and the Indian Registration (Bombay Amendment) Act, 1939 (Bom. XI of 1939).

Indian Registration (Bombay Amendment) Act, 1947 (Bom. XIV of 1947).

Indian Registration (Bombay Provision for Uniformity and Amendment) Act, 1958 (Bom. XXXV of 1958).

Indian Registration (Maharashtra Amendment) Act, 1960 (Mah. XIX of 1960).

Registration (Bombay Amendment) See Bombay Public Trusts (Unification and Amendment) Act, 1959 (Bom. VI of 1960).

Registration (Maharashtra Amendment) Act, 1966 (Mah. XXXV of 1966).

Registration (Maharashtra Amendment) See Bombay Public Trusts (Amendment) Act, 1970 (Mah. XX of 1971).

Registration (Maharashtra Amendment) Act, 1974 (Mah. XXIX of 1974).

Registration (Maharashtra Amendment) Act, 1975 (Mah. XLIX of 1975).

REGISTRATION OF MARRIAGES—

See "Husband and Wife".

REGISTRATION OF SOCIETIES—

See "Societies".

REMOVAL OF DISQUALIFICATIONS—

See "Disqualifications—Removal of".

RENT CONTROL—

Maharashtra Rent Control Act, 1999 (Mah. XVIII of 2000).

REPEALING AND AMENDING ACT—

See "Laws".

Abolition of Cash Grants-See "Land Tenures".

Abolition of Inam-See "Land Tenures".

Building-See "Building".

Cotton Contracts-See "Contract".

C. P. and Berar-Finance-See "Finance".

State Guarantees-See "Guarantees".

State Aid to Industry-See "Industry."

Grant-in-Aid (C. P. and Berar)-See "Local Authorities".

Baronetcy-See "Private Property" and "Trust and Trustees."

Lepers-See "Public Health".

Public Service Selection-See "Public Service".

Rehabilitation Loan (C. P. and Berar)-See "Refugees".

REVENUE—

Bombay Abkari Act, 1878 (Bom. V of 1878).

Bombay Abkari (Amendment) Act, 1912 (Bom. XII of 1912).

Excise Act, 1915 (C. P. and Berar) (II of 1915).

Maharashtra Entertainments Duty Act (I of 1923).

Indian Tariff (Bombay Application) Regulation Act, 1943 (Bom. Reg. I of 1943).

Revocation of Land Revenue Exemptions Act, 1948 (XXXVII of 1948).

Salsette Estate (Land Revenue Exemption Abolition) Act, 1951 (Bom. XLVII of 1951).

Merged Territories Revenue Laws (Exercise of Powers) Act, 1951 (M. P.) (XX of 1951).

Land (Special Assessment) Hyderabad Act, 1952 (XXXII of 1952).

Recovery of Arrears of Land Revenue and Rent Act, 1953 (M. P.) (V of 1953).

Maharashtra Motor Vehicles Tax Act (LXV of 1958).

Maharashtra Lotteries (Control and Tax) and Prize Competitions (Tax) Act (LXXXII of 1958).

Maharashtra Electricity Duty Act (XXVI of 2016).

Maharashtra Motor Vehicles (Taxation of Passengers) Act (LXVII of 1958).

Bombay Motor Vehicles (Taxation of Passengers) (Amendment and Validation) Act, 1981 (Mah. IV of 1982).

Maharashtra Stamp Act (LX of 1958).

Bombay Tobacco Laws (Repeal) Act, 1959 (Bom. LXXIII of 1959).

Maharashtra Municipal Property-Tax Board Act, 2011 (Mah. XIV of 2011).

Maharashtra Agricultural Income-Tax Act, 1962 (Mah. XLI of 1962).

Maharashtra Provisional Collection of Taxes Act, 1963 (Mah. V of 1963).

Maharashtra Tax on Sale of Electricity Act, 1963 (Mah. XXI of 1963).

Maharashtra Land Revenue Code, 1966 (Mah. XLI of 1966).

Maharashtra Land Revenue Code (Amalgamation of Bombay and Konkan Division)

Act, 1983 (Mah. XXXIX of 1983).

Maharashtra Tenancy Law and Maharashtra Land Revenue Code (Amendment and Validation of Appointments of Certain Officers and Proceedings) Act, 1981 (Mah. V of 1982).

Stamps (Amendment) Act, 1971 (Mah. L of 1971).

Maharashtra Temporary Increase in Taxes on Motor Vehicles and Passengers Act, 1972 (Mah. XI of 1972).

Maharashtra Land Revenue (Revival of Certain Rules relating to Non-agricultural Assessment) Act, 1972 (Mah. XXV of 1972).

Maharashtra Temporary Increase in Entertainments Duty and Education Cess Act, 1972 (Mah. VII of 1972).

Maharashtra Special Provision for Payment of Stamp Duty Act, 1974 (Mah. XXXIII of 1974).

Maharashtra Special Provision for Payment of Stamp Duty (No. 2) Act, 1974 (Mah. LII of 1974).

Maharashtra Increase of Land Revenue and Special Assessment Act, 1974 (Mah. XX of 1974).

Maharashtra State Tax on Profession, Trades, Callings and Employments Act, 1975 (Mah. XVI of 1975).

Tax on Buildings (With Larger Residential Premises) (Re-enacted) Act, 1979 (Mah. XXIX of 1979).

Maharashtra Sinhastha Fair Pilgrim Tax Act, 1980 (Mah. II of 1980).

See "Luxury-cum-Entertainment".

Maharashtra Value Added Tax Act, 2002 (Mah. IX of 2005).

REVENUE TRIBUNAL—

See " Courts ".

ROPEWAYS, AERIAL—

See "Communications".

SALARIES, ALLOWANCES AND PENSION—

Maharashtra Legislature Members' Salaries and Allowances Act (XLIX of 1956).

Maharashtra Ministers' Salaries and Allowances Act (XLVIII of 1956).

Maharashtra Legislature Council (Chairman and Deputy Chairman) and the Maharashtra Legislative Assembly (Speaker and Deputy Speaker) Salaries and Allowances Act (XLVII of 1956).

Leaders of Opposition in Maharashtra Legislature (Salaries and Allowances)

Act, 1978 (Mah. VIII of 1978).

Legislature Members' Pension Act, 1977 (Mah. I of 1977).

SALES TAX—

See "Revenue".

SEALS—

Maharashtra Public Authority Seals Act, 1883 (V of 1883).

Bombay Seals Act (Application to Scheduled Areas) Regulation, 1951 (Bom. Reg. I of 1951).

SECONDARY SCHOOLS CERTIFICATE EXAMINATION—

See " Education".

SECURITY MEASURES—

The Maharashtra Private Security Guards (Regulation of Employment and Welfare) Act, 1981 (Mah. LVIII of 1981).

The Maharashtra State Security Corporation Act, 2010 (Mah. VI of 2010).

The Maharashtra Suraksha Dal Act. 1987 (Mah. Ill of 1987).

SEPARATION OF JUDICIAL AND EXECUTIVE FUNCTIONS—

Bombay Separation of Judicial and Executive Functions Act, 1951 (Bom. XXIII of 1951).

Bombay Separation of Judicial and Executive Functions (Supplementary) Act, 1954 (Bom. VIII of 1954).

Bombay Separation of Judicial and Executive Functions (Extension) and the Code of Criminal Procedure (Provisions for Uniformity) Act, 1958 (Bom. XCVII of 1958).

SHOPS AND ESTABLISHMENTS—

See "Trade".

SHORE NUISANCES—

See " Nuisances."

SMALL CAUSE COURTS—

See " Courts ".

SOCIETIES—

Societies Registration (Amendment) Act, 1912 (Bom. II of 1912).

Co-operative Land Mortgage Banks Act, 1937 (C. P. and Berar) (I of 1937).

Co-operative Societies (Amendment of Liquidators Orders) Validation Act, 1945 (C. P. and Berar) (X of 1945).

Societies Registration (Bombay Amendment) Act, 1948 (Bom. LIII of 1948).

Societies Registration (Bombay Amendment) Act, 1956 (Bom. XI of 1956).

(G.C.P.) H 1077—9 (2043-12-2019)

Co-operative Societies (Reconstitution and Formation) Act, 1956 (M. P. XVIII of 1956).

Societies Registration (Bombay Extension and Amendment) Act, 1958 (Bom. LXXVI of 1958).

Maharashtra Co-operative Societies Act, 1960 (Mah. XXIV of 1961).

Societies Registration (Maharashtra Amendment) Act, 1968 (Mah. XI of 1968).

Maharashtra Specified Co-operative Societies (Postponement of Elections due to Scarcity Conditions in the State) Act, 1972 (Mah. XLV of 1972).

Maharashtra Co-operative Sugar Factories Under Erection (Postponement of Elections) Act, 1974 (Mah. LIX of 1974).

Maharashtra Specified Co-operative Societies (Postponement of Elections due to Emergency) Act, 1975 (Mah. XLIII of 1975).

Maharashtra Specified Co-operative Societies (Postponement of Elections due to draught conditions in the State) Act, 1983 (Mah. XXVIII of 1983).

Maharashtra Co-operative Sugar Factories (under erection and certain others) (Postponement of Election) Act, 1977 (Mah. XLIX of 1977).

Maharashtra Co-operative Societies (Amendment and Validation) Act, 1977 (Mah. LVIII of 1977).

SPECIAL TRIBUNAL—

See "Courts".

SPECIAL SUIT AND PROCEEDINGS—

See "Civil Procedure and Practice."

SPECIFIED COMMODITIES—

See "Markets and Fairs" or "Local Authorities".

STAMP—

See "Revenue."

Maharashtra State Commission for Safai Karmacharis Act, 1997 (Mah. XLV of 1997).

STATE COMMISSION—

Maharashtra Rajiv Gandhi Science and Technology Commission Act, 2004 (Mah. XV of 2004).

Maharashtra State Minorities Commission Act, 2004 (Mah. XXXI of 2005).

Maharashtra State Commission for Backward Classes Act, 2005 (Mah. XXXIV of 2006).

STATE GUARANTEES—

See "Guarantees".

STATE RESERVE POLICE—

See "Police".

STATISTICS—

See "Local Authorities".

STEAM VESSELS—

See "Transport".

SUIT—

See "Civil Procedure and Practice".

SUGAR-CANE CESS—

See "Agriculture".

See "Revenue".

SURVEY—

Land Survey Act, 1947 (C. P. and Berar) (XLII of 1947).

TARIFF—

See "Revenue".

TAX—

Tax on Sale of Forest Produce-See "Forest".

Tax on Sale of Electricity-See "Revenue".

Tax on Luxury-See "Hotels and Lodging".

Tax on Agricultural Income-See "Revenue".

Tax on Sales-See "Revenue".

Tax on Transfer of Right to use-See "Revenue".

Tax on Building and Lands-See "Revenue".

Tax on Motor Vehicles-See "Revenue".

Tax on Passengers-See "Revenue".

Tax on Pilgrims-See "Revenue".

Tax on Residential Premises-See "Revenue".

Tax on Casinos-See "Revenue".

Tax on Profession-See "Revenue".

Maharashtra Taxation Laws Offences (Extension of Period of Limitation) Act, 1982 (Mah. XXII of 1982).

MOTOR VEHICLES—

See "Revenue".

PROVISIONAL COLLECTION OF—

See "Revenue".

LOTTERIES AND PRIZE COMPETITIONS—

See "Revenue".

LUXURY-Cum-ENTERTAINMENT—

Maharashtra Luxury-cum-Entertainment and Amusement tax on Holders of Television Sets (Repeal) Act, 1985 (Mah. V of 1985).

SALES AND PURCHASES—

See "Revenue".

INCOME-TAX, AGRICULTURAL—

See "Revenue".

PILGRIM TAX—

See "Revenue".

H 1077—9a

SUGAR-CANE—

See "Revenue".

Maharashtra Regulation of Sugar-cane Price (Supplied to Factories) Act, 2013 (Mah. XXXIII of 2013).

TEMPLES—

Pandharpur Temples Act, 1973 (Mah. IX of 1974).

Shree Siddhi Vinayak Ganpati Temple Trust (Prabhadevi) Act, 1980 (Mah. VI of 1980).

Eradication of Unfair Activities at Pandharpur Temples (for providing better facilities for worship) Act, 1980 (Mah. XI of 1980).

Maharashtra Shree Sai Baba Sansthan Trust (Shirdi) Act, 2004 (Mah. XIV of 2004).

TENANCY—

See "Agriculture."

TITLES—

The Maharashtra (Change of Short Titles of Certain Bombay Acts) Act, 1980 (Mah. XV of 1980).

The Maharashtra (Change of Short Titles of Certain Bombay Acts) Act, 2011 (Mah. XXIV of 2012).

TOLLS-

See "Revenue."

TOWN PLANNING—

Bombay Town Planning Schemes Bombay City Nos. II, III and IV (Mahim Area) Validation Act, 1956 (Bom. Ill of 1957).

Maharashtra Regional and Town Planning Act, 1966 (Mah. XXXVII of 1966).

Maharashtra Metropolitan Planning Committees (Constitution and Functions) (Continuance of Provisions) Act, 1999 (Mah. V of 2000).

Maharashtra District Planning Committees (Constitution and Functions) Act, 1998 (Mah. XXIV of 1998).

TRADE—

Markets and Fairs Act, 1862 (Bom. IV of 1862).

Shops and Establishments (Regulation of Employment and Conditions of Service) Act, Maharashtra (LXI of 2017).

Bombay Essential Commodities and Cattle Control Act, 1958 (Bom. LXII of 1958).

Maharashtra Warehouses Act (V of 1960).

Maharashtra Raw Cotton (Procurement, Processing and Marketing) Act, 1971 (Mah. XLVII of 1971).

Essential Commodites (Maharashtra Amendment) Act, 1976 (Mah. I of 1976).

TRADE DISPUTES—

See "Labour".

TRAMWAYS—

See "Transport."

TRANSFER OF PROPERTY—

Transfer of Property and the Indian Registration (Bombay Amendment) Act, 1939 (Bom. XIV of 1939).

Transfer of Property (Bombay Provision for Uniformity and Amendment) Act, 1959 (Bom. LVII of 1959).

Transfer of Property in Goods—Sales Tax on....See "Revenue".

TRANSFER OF RIGHTS—

To use any goods-Sales Tax on....See "Revenue".

TRANSPORT—

Maharashtra Steam Vessels Act (II of 1864).

Maharashtra Ferries and Inland Vessels Act (II of 1868).

Maharashtra Public Conveyances Act (VII of 1920).

Indian Tramways (Bombay Amendment) Act, 1948 (Bom. L of 1948).

Indian Ports (Maharashtra Amendment) Act, 1962 (Mah. IV of 1963).

Bombay Landing and Wharfage Fees and Indian Ports (Amendment) Act, 1972 (Mah. VIII of 1973).

Motor Vehicles (Maharashtra Amendment) Act, 1972 (Mah. X of 1972).

Motor Vehicles (Maharashtra Second Amendment) Act, 1972 (Mah. VI of 1973).

Motor Vehicles (Maharashtra Amendment) Act, 1975 (Mah. XIX of 1975).

Motor Vehicles (Maharashtra Amendment) Act, 1976 (Mah. XLVII of 1976).

Maharashtra Requisitioning (and Control) of Motor Vehicles Act, 1965 (Mah. II of 1966).

Maharashtra Truck Terminal (Regulation of Location) Act, 1997 (Mah. XXXI of 1997).

TREASURE-TROVE—

Indian Treasure-Trove (Extention to the Hyderabad and Saurashtra Areas of Bombay State) Act, 1957 (Bom. XXXIII of 1958).

TREES—

Maharashtra Felling of Trees (Regulation) Act, 1964 (Mah. XXXIV of 1964).

Maharashtra Sale of Trees by Occupants belonging to Scheduled Tribes (Regulation) Act, 1969 (Mah. XXIII of 1969).

Maharashtra (Urban Areas) Preservation of Trees Act, 1975 (Mah. XLIV of 1975).

TRIBALS—

Tribal Areas Regulation, 1359 F (Hyderabad) (III of 1359 F).

Maharashtra Tribals Economic Condition (Improvement) Act, 1976 (Mah. V of 1977).

TRUSTS AND TRUSTEES—

Maharashtra Public Trusts Act (XXIX of 1950).

Madhya Pradesh Public Trusts (Amendment) and Validation Act, 1958 (Bom. XLIX of 1958).

Bombay Public Trusts (Unification and Amendment) Act, 1959 (Bom. VI of 1960).

Sir Currimbhoy Ebrahim Baronetcy (Repeal and Distribution of Trust Properties) Act, 1959 (Bom. IX of 1960).

Sir Cowasjee Jahangir Baronetcy (Repealing) Act, 1964 (Mah. XXVI of 1965).

Maharashtra Baronetcies (Amendment) Act, 1974 (Mah. XXVIII of 1974).

Public Trusts (Validating Provisions and Amendment) Act, 1977 (Mah. XXXV of 1977).

Administrator General and Official Trustees (Maharashtra Amendment) Act, 1982 (Mah.

XIV of 1983).

TULJAPUR—

Tuljapur Development Authority Act, 2008 (Mah. Ill of 2009).

UNCLAIMED BODIES—

Unclaimed Property Act, 1333 F (Hyderabad) (I of 1333 F).

Maharashtra Anatomy Act (XI of 1949).

UNIVERSITIES—

Maharashtra Agricultural Universities (Krishi Vidyapeeths) Act, 1983 (Mah. XLI of 1983).

Maharashtra Prevention of Malpractices at University, Boards and Other Specified Examinations Act, 1982 (Mah. XXXI of 1982).

Dr. Babasaheb Ambedkar Technological University Act, 2014 (Mah. XXIX of 2014).

The Yashwantrao Chavan Maharashtra Open University Act, 1989 (Mah. XX of 1989).

Bombay University (Temporary Postponement of Election of University Representatives) Act, 1989 (Mah. XL of 1989).

Maharashtra Public Universities Act, 2016 (Mah. VI of 2017).

Maharashtra Animal and Fishery Sciences University Act, 1998 (Mah. XVIII of 1998).

The Kavi Kuluguru Kalidas Sanskrit Vishvavidyalaya (University) Act, 1997 (Mah. XXXIII of 1997).

Maharashtra University of Health Sciences Act, 1998 (Mah. X of 1999).

Amity University Act, 2014 (Mah. XIII of 2014).

Dr. Ajeenkya D. Y. Patil University Act, 2014 (Mah. Ill of 2015).

FLAME University Act, 2014 (Mah. II of 2015).

National Law University Act, 2014 (Mah. VI of 2014).

Spicer Adventist University Act, 2014 (Mah. XIV of 2014).

VETERINARY PRACTITIONERS—

See "Medical Profession".

VILLAGE PANCHAYATS—

See "Local Authorities".

WAGERS-

Bombay Act for Avoiding Wagers (Amendment) Act, 1865 (Bom.III of 1865).

WAREHOUSES-

See "Trade".

WATAN—

See "Land Tenure".

WATER CONSERVATION—

Maharashtra Water Conservation Corporation Act, 2000 (Mah. Ill of 2001).

WATER POLLUTION—

See "Pollution".

WATER RESOURCES REGULATORY AUTHORITY—

Maharashtra Water Resources Regulatory Authority Act, 2005 (Mah. XVIII of 2005).

WATER SUPPLY—

Maharashtra Jeevan Authority Act, 1976 (Mah. XLVIII of 1976).

Maharashtra Drinking Water Supply Requisition Act, 1983 (Mah. XX of 1983).

WELFARE FUND—

See "Labour".

Maharashtra Advocates Welfare Fund Act, 1981 (Mah. LXI of 1981).

Maharashtra Private Security Guards (Regulation of Employment and Welfare) Act, 1981 (Mah. LVIII of 1981).

WHIPPING, ABOLITION OF—

See "Criminal Law and Procedure".

WILLS—

Wills and Intestacy Regulation (Sections 4, 5, 6 and 7) (Bengal), (M. P.) 1799 (V of 1799).

WOMEN'S RIGHTS—

See "Personal Law".

GOVERNMENT CENTRAL PRESS, MUMBAI

Maharashtra Government Publication can be obtained from—

• THE DIRECTOR

GOVERNMENT PRINTING, STATIONERY AND PUBLICATION

Maharashtra State

Netaji Subhash Road,

MUMBAI 400 004.

Phone: 022 - 23632693, 23630695, 23631148, 23634049

THE MANAGER

Government Photozinco Press and Book Depot

Photozinco Press Area, Near G.P.O.,

PUNE 411 001.

Phone: 020 - 26125808, 2025822, 26128920

• THE MANAGER

GOVERNMENT PRESS AND BOOK DEPOT Civil Lines,

NAGPUR 440 001.

Phone: 0712 - 2562615, 2563597

• THE ASSTT. DIRECTOR

GOVERNMENT STATIONERY, STORE AND BOOK DEPOT

Shaha Ganj, Near Gandhi Chowk,

AURANGABAD 431 001.

Phone: 0240 - 2331468

• THE MANAGER

GOVERNMENT PRESS AND STATIONERY STORE

Tarabai Park,

KOLHAPUR 416 003.

Phone: 0231-2650395, 2650402

AND THE RECOGNISED BOOKSELLERS